Abortion and Female Power in the Household Evidence from Labor Supply Sonia Oreffice NOTA DI LAVORO 41.2003

APRIL 2003 KNOW – Knowledge, Technology, Human Capital

Sonia Oreffice, The University of Chicago

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/web/activ/_wp.html

Social Science Research Network Electronic Paper Collection: http://papers.ssrn.com/abstract_id=XXXXXX

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei

Abortion and Female Power in the Household Evidence from Labor Supply

Summary

Legalization of abortion in the 1970s represents a major cultural change: it gives women a higher degree of freedom to directly control their fertility, allowing them to ultimately decide upon children without man's consent and to decrease uncertainty in their expected labor market returns.

This public policy and its implications on female behavior have been so far analyzed through its direct consequences on fertility and fertility technology, primarily on women actually experiencing an abortion.

However, it seems relevant to evaluate its indirect effects on female bargaining power within the household, for all women that face abortion as an actual opportunity, without necessarily experiencing one. Although indirect, this impact may be widespread since in principle it concerns all couples where the woman is in her fertile age.

I focus on the indirect effect of abortion legalization in the United States on women's position in the household. My findings suggest that the legalization positively affected female bargaining power.

In a dynamic model of possible commitment to marital contract, I analyze households that were already formed before the legalization and I test for the renegotiation of the household's initial contract in the presence of the shock to bargaining power balance represented by abortion legalization. Results seem to reject the full commitment assumption.

Specifically, I analyze the legalization of abortion as a distribution factor in a collective model of household behaviour through its effects on female labor supply and find that (controlling for the number of children) the legalization significantly decreased labor supply of married women in their fertile age and significantly increased their husbands', while no such effect is found for older married women nor for single and divorced women or men.

Keywords: Abortion, collective household behavior, labor supply

JEL: D12, J12, J22, K39

The author would like to thank Gary Becker, Marianne Bertrand, Pierre-André Chiappori, Erik Hurst, Steve Levitt and participants to the EEA 2002 Conference, Chiappori-Townsend working group for their helpful insights and suggestions. The author is also very grateful to Carlo Carraro and Gino Cateau. Errors are the author's.

Address for correspondence: Sonia Oreffice The University of Chicago Department of Economics 5801 South Ellis Ave. Chicago, IL 60637 USA E-mail: soreffic@midway.uchicago.edu

I. Introduction

Abortion legalization has always been one of the most debated policy issues in the United States and in many Western countries, recently also at the United Nations level². In the United States in particular, there have been many federal and state laws, amendments and regulations that have alternatively modified the conditions under which abortion is legal and the restrictions to women's autonomous decision.

Legalization of abortion undoubtedly affects women's freedom to ultimately control their fertility. It represents a new right to female autonomy in the fertility decision, an issue that in principle concerns both spouses involved in a relationship.

It is interesting to investigate and evaluate the impact of abortion legalization on female empowerment, specifically within the household, through its indirect effects on spouses' bargaining power. Such a right to women's autonomy in planning fertility may increase the female intra-household power and influence the household allocation of resources through two main channels. First, this right represents a new opportunity for the woman to decide upon children with or without man's consent, possibly affecting his fertility choices. It may be the case that spouses disagree on the number and timing of children, so that this new availability gives women more freedom with respect to men, possibly binding men's fertility decisions³. Second, this opportunity to avoid unwanted pregnancies decreases women's uncertainty about their future career which, ex ante, is likely to make their expected life-time earnings larger than before the legalization. Abortion statistics (Allan Guttmacher Institute, US, 1970s) show that the practice of abortion is indeed seen as an opportunity by women involved in a stable relationship: around 27 % of women undergoing an abortion are legally married⁴. Hence, abortion may represent a credible threat and bound for men. Also, the vast majority of women undergoing an abortion are reported to do so because they do not want the child or they want to delay fertility, suggesting that this opportunity is used when an actual voluntary choice is involved, not merely for extreme contingencies such as rape, incest, fetal or woman's health problems (precisely, about 90 % for "unwantedness" reasons

² Refer, for instance, to the declarations at the United Nations World Conference on Women at Beijing (1995) and Conference on Population and Development in Cairo (1994).

³ In the literature on reproduction and fertility goals, potential disagreement between spouses is accounted for. For example, Coombs et al. (1979) state that "reproductive goals of one marital partner cannot with confidence be assumed to represent the views of the other". Thomson et al. (1990) show that "disagreeing couples experienced fertility rates midway between couples who wanted the same smaller or larger number of children". In particular, Williams (1994) says that a substantial proportion of births that are unwanted by mothers are wanted by fathers.

⁴ Of all women undergoing an abortion in the 1970s, 61 % are between the ages of 25 and 40, age group on which I focus for my empirical analysis. See the data description section for details.

versus 10 % for those contingencies)⁵. Moreover, sociological and gender studies show that abortion legalization is regarded specifically as a right which allowed women to decrease male domination within families⁶.

These statistical and literature evidence seems to suggest that the impact of abortion legalization on female bargaining power in the household may be economically sizable and interesting to model and estimate.

Actually, legalization of abortion leads to many changes, as it is acknowledged in the literature. We can distinguish among three main effects of this phenomenon.

- Fertility effect: direct impact. The legalization of abortion straightforwardly allows women to get an abortion: through their decision to abort, the legalization affects demographic variables, with a substantial decline in fertility. The literature (Levitt Donohue (2001), Levine et al. (1996)) shows that this direct impact on fertility subsequently led to changes in teenagers' behavior on fertility, "out of wedlock" births and in crime.
- 2. Technology effect: indirect impact. The availability of abortion indirectly affects the fertility technology because it represents a sure mean of ultimately avoiding a pregnancy, as, for instance, the contraceptive pill is. Decisions on fertility, marriage and human capital investments may thus be affected, as it is precisely shown in Goldin-Katz (2001), Angrist-Evans (1996). These papers show that there is an indirect effect on long run increases in labor market participation and education of women who conceivably were on the contraceptive pill or took advantage of abortion legalization, not necessarily distinguishing across marital status.
- 3. Bargaining power effect: I propose a new analysis dimension, the indirect impact within families on household's decision process. The legalization of abortion represents an opportunity for women to decide upon their fertility. This may increase female bargaining power within the household and thus influence the observed households' optimal allocations, such as labor supply, participation and any specific consumptions variables that are susceptible to be influenced by bargaining power changes.

Most of the literature on abortion or reproductive technology concentrates in either one or both of the first two impacts. However, the bargaining power effect may be relevant and widespread since it potentially affects all couples possibly dealing with fertility decisions⁷: typically all couples where the woman is in her fertile age. In fact, all women for whom abortion becomes an available opportunity to self-determine fertility, including those that eventually never experience

⁵ These statistics on the reasons why abortion was undertaken are from 1987: they were not available before. However, in these past fifteen years unwantedness seems to represent the main motivation, so that these percentages are likely to reflect the 1970s as well.

⁶ Historically, those studies were developed even before the literature regarding abortion as a source of female empowerment in society. Now this issue is taken further by considering abortion as one of women's reproductive and sexual rights.

⁷ My analysis does not apply to lesbian or homosexual couples nor on the heterosexual couples where one spouse is totally dominant in the relationship. I refer to heterosexual couples where there is room for potential renegotiation of bargaining power balance between a man and a woman.

an abortion, may increase their intra-household bargaining power towards their spouses as a consequence of its legalization. Not only their power should increase, but correspondingly their male spouses' power should decrease. Also, this impact should show up immediately in the cohorts of women exposed to the legalization, since the legalization per se, as a new right to female autonomy, changes the bargaining power balance. Therefore, even though this intra-household effect is more indirect than the first order fertility decline effect, it may potentially affects a broader segment of the population. Then, the overall consequences of abortion legalization may be of much greater proportion than previously acknowledged. In addition to this, if we specifically consider an intertemporal framework, the bargaining power effect allows to investigate two further related issues: 1) to what extent spouses renegotiate their marital⁸ contract when there is such an unanticipated shock to their bargaining power balance; 2) how different marital contracts of couples who got married before the legalization are from those of couples who got married after the legalization. In this present paper I do not address the second issue, mainly because of small sample size⁹.

In order to test and measure the bargaining power effect of abortion legalization, I will focus on labor supply decisions, as in the literature on bargaining power (Chiappori (1992), Chiappori et al. (2002)). The intuition for how an increase in female intra-household bargaining power due to abortion legalization can affect spouses' labor supply is the following: ceteris paribus, an increase in women's bargaining power gives women a higher share of household resources and this translates into a positive income effect for them; if the income effect goes in the regular direction (if leisure is a normal good), women's labor supply decreases as a consequence of legalization of abortion.

It is crucial to emphasize that this labor supply prediction of the bargaining power effect goes in the opposite direction with respect to the fertility and the technology effects found in most of the literature. Indeed, if women get access to better fertility technology and fertility itself declines, becoming less of a constraint, female labor supply increases, as it is well documented in the literature on abortion and birth control such as the contraceptive pill. Moreover, transformations in fertility and its technology can be dealt with both a unitary and a non-unitary setting of household behavior, whereas the impact on intra-household bargaining power can be analyzed only in a non-unitary framework. In fact, while the first two effects can influence women regardless of the marital status, the latter impact on bargaining power may affect only those involved in a relationship with a male spouse, i.e. legally married or cohabiting women. This male spouse is also predicted to be affected by the legalization.

⁸ Throughout all my analysis, I use the word married as synonym for both legally married and cohabitants. By the same token, all the related words such as husband, wife, marriage, marital should be interpreted as referring also to cohabitants.

⁹ In my current data set, only less than ¹/₄ of couples got married after the legalization. However, I am planning to switch to CPS data to try to analyze this further question.

The collective model of household behavior provides a natural theoretical framework to perform the analysis of this bargaining effect. Each spouse is assumed to have his own utility to maximize and spouses are assumed to take pareto-efficient decisions. The legalization of abortion may modify the household's environment and spousal opportunities, affecting the intra-household decision process and optimal allocations only through its impact on spousal bargaining within the household. In a dynamic setting, I focus on couples who got married before the legalization of abortion in order to avoid marriage market selection bias issues, and essentially to test for the renegotiation of their marital contracts due to the legalization.

I consider both the national legalization of abortion in 1973 and the early state legalizations by California, New York state, Washington state, Alaska and Hawaii in 1970 as sources of variation in the availability of abortion¹⁰. In fact these laws represented a very "open" legalization: women were allowed to decide autonomously (within a specified period from conception) and psychological distress and unwantedness were contemplated as motivations to seek an abortion¹¹. I use the Panel Study of Income Dynamics (PSID) from 1970 to 1979, with household data across the United States, to assess the impact of abortion on female bargaining power.

I estimate the impact of abortion legalization on married couples, singles and divorced. I find that the legalization significantly decreased labor supply of married women in their fertile age and significantly increased their husbands', whereas for married women beyond the fertile period, for singles, divorced women and men there is no such effect. My results suggest that couples indeed renegotiate their marital contract in response to abortion legalization, when it represents an actual opportunity (fertile women). Furthermore, this renegotiation leads to an increase in these women's intra-household bargaining power relative to their spouse.

Section II presents the theoretical framework to analyze the impact of abortion legalization on female bargaining power in the household and the possible marital contract renegotiation, namely an intertemporal collective model of labor supply. The following Section III provides the labor supply specifications. Section IV contains a description of the data and Section V presents the empirical results. The last sections Section VI and VII discuss sensitivity analysis and alternative explanations and Section VIII concludes the paper.

¹⁰ I do not consider twelve other states that inacted abortion laws between 1967 and 1970 because they legalized abortion only under special restrictive circumstances (life danger, rape). These states laws are also generally discarded in the abortion literature.

¹¹ See the appendix for a detailed timing and description of these laws.

II. Intertemporal collective model of labor supply with possible marital contract renegotiation

In a collective household behavior setting, I consider dynamic models of labor supply, either with or without commitment towards marital contract.

Each spouse in the household has a distinct utility function and the decision process is pareto-efficient. The household is composed of two decision makers, called husband and wife¹². They are assumed to be infinitely lived.

Let h_t^i and C_t^i for i=1,2 denote respectively member i's labor supply and consumption of a private composite good in period t; leisure is assumed to be a normal good. Let b_t denote the household net savings on a risk-free asset in period t. Let's assume the standard specification of individuals with egoistic preferences, where welfare of member i does not depend on consumption or leisure of the other member (no externality, no public good): $U^i(1-h_t^i, C_t^i, z)$ where U is strictly quasi-concave, increasing, continuously differentiable and z are preferences parameters. From now on, i = 1= wife; i = 2 = husband.

Spouses negotiate their marital contract and decide their optimal allocations of savings, consumption and leisure at the time of marriage in t = 0. Whether they decide upon the optimal allocations for the whole time horizon, and whether their observed behavior is possibly affected by abortion legalization, depends on the degree of commitment¹³.

1) Full commitment

Under full commitment, spouses stick to their marital contract throughout the whole period, even though the variables influencing bargaining power between spouses may change over time due to the legalization of abortion. The behavior of savings, labor supply and consumption over time is not affected by changes in these variables happening after the marriage.

In fact, the household intertemporal efficient allocation of resources can be written as the solution of the following program:

¹² I focus on spouses where the wife is in her fertile age, since abortion legalization may increase female bargaining power only for couples where fertility decisions are still an issue at stake. Once again, husband and wife stand for both legally married and cohabitant spouses.

¹³ I may use the same theoretical commitment framework to think of fertility decisions in terms of number and timing of children. They would be decided ex ante at the time of marriage and possibly renegotiated in favor of women at the time of legalization of abortion. In the data, preferences on number and timing of children can be captured by preference parameters and by the variables number of children and presence of infants in the household.

$$\max_{\{h_{t}^{i}, C_{t}^{i}\}_{i=1,2;t=0,\infty}} \sum_{t=0}^{\infty} \beta^{1} U^{1} (1 - h^{1}_{t}, C_{t}^{1}, z)$$
st:

$$\sum_{t=0}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z) \ge U^{2^{0}}$$

$$\forall t \rightarrow p_{t} (C_{t}^{1} + C_{t}^{2}) + b_{t} \le w_{t}^{1} h_{t}^{1} + w_{t}^{2} h_{t}^{2} + y_{t} + b_{t-1}$$

$$0 \le h_{t}^{1} \le 1; 0 \le h_{t}^{2} \le 1$$

$$\forall t \in [0, +\infty)$$

$$\Rightarrow C_{t}^{i} = C^{i} (p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, U^{2^{0}})$$

$$\Rightarrow h_{t}^{i} = h^{i} (p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, U^{2^{0}})$$

$$\Rightarrow b_{t} = b(p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \bar{U^{20}})$$

 $\forall i = 1.2$

where y is the household non-labor income, w is the wage rate, r is the interest rate at which they can lend or borrow

money, \overline{U}^2 is the husband's reservation utility at the time of marriage; discount factors β^1 and β^2 and the interest rate r are assumed to be constant over time without loss of generality and, together with preference parameters, are omitted from any optimal allocation for notational simplicity (in particular, I assume r = 0). In general, the value of the reservation utility is a function of all information available at the time of the decisions. This information set is composed of all possible variables affecting bargaing power; in particular, I focus on distribution factors s, which consist of all other variables that modify household's environment and spousal opportunities without changing preferences nor the pareto-set; specifically, I analyze legalization of abortion as an additional opportunity to the wife to directly control her fertility, possibly increasing her bargaining power. In a full commitment setting, I denote the

reservation utility as U^{2^0} because optimal decisions are taken at the time of marriage t = 0.

The optimal allocation functions $C^{i}(.), h^{i}(.), b()$ and the reservation utility function are assumed to be stationary. Also, the reservation utility, as well as the optimal allocations, is assumed not to depend on past realizations of nonlabor income and wages for notational simplicity.

Theorem. Under full commitment, the above program is equivalent to the existence of the pareto-weight μ^0 , constant throughout the whole period, so that spouses solve the following optimization:

$$\begin{aligned} \max_{\{h_t^i, C_t^i\}_{t=1,2;t=0,\infty}} \mu^0 \sum_{t=0}^{\infty} \beta^1 U^1 (1 - h^1_t, C_t^1, z) + (1 - \mu^0) \sum_{t=0}^{\infty} \beta^2 U^2 (1 - h_t^2, C_t^2, z) \\ st : \forall t \to p_t (C_t^1 + C_t^2) + b_t \le w_t^1 h_t^1 + w_t^2 h_t^2 + y_t + b_{t-1} \\ 0 \le h_t^1 \le 1; 0 \le h_t^2 \le 1 \\ \forall t \in [0, +\infty) \end{aligned}$$
$$\Rightarrow C_t^i = C^i (p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^0) \\ \Rightarrow h_t^i = h^i (p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^0) \\ \forall i = 1, 2 \\ \Rightarrow b_t = b(p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^0) \end{aligned}$$

Proof. Lagrange theorem on the existence of Lagrange multipliers.

With full commitment, only the distribution of bargaining power at t = 0 matters: therefore, μ^0 is affected by bargaining power variables only of time t = 0. Since spouses commit ex ante to the stream of optimal allocations, legalization of abortion does not have any impact on households who got married before the legalization. Even though it may represent a change in distribution factors, spouses keep the power balance they established in t = 0. Therefore, under full commitment, abortion legalization should not have any impact on household behavior. In fact, the full commitment assumption does not allow to identify the reason why there is no such impact: it may be the case that abortion legalization does not affect bargaining power at all (it is not a distribution factor) or that indeed it represents a distribution factor but couples already married before the legalization are in fact fully committed.

The full commitment assumption may be too restrictive: it may not be realistic to assume that spouses commit to allocations decided at the time of marriage for their entire life together. In terms of the model, spouses may change their bargaining power balance and renegotiate their marital contract. Thus, the pareto-weight may vary accordingly over time. We can consider two sub-models of the limited commitment, both allowing for renegotiation of the marital contract due to unanticipated changes in bargaining power variables: a model of commitment but with full renegotiation at the time of the unexpected shock; a model without commitment where spouses negotiate their optimal allocations at any date.

2) Commitment with renegotiation due to unanticipated shocks to bargaining power

Under commitment with renegotiation, spouses stick to their marital contract until there is an unanticipated shock to their bargaining power, namely the legalization of abortion, making them renegotiate their optimal allocations at the time of the shock. The behavior of savings, labor supply and consumption over time possibly changes after the marriage in response to abortion legalization.

In fact, the household intertemporal efficient allocations of resources can be written as the solutions of the following program:

if
$$t < legal$$

$$\max_{\{h_{t}^{i}, C_{t}^{i}\}_{i=1,2;t=0,\infty}} \sum_{t=0}^{\infty} \beta^{1} U^{1} (1 - h_{t}^{1}, C_{t}^{1}, z)$$

$$st : \sum_{t=0}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z) \ge U^{20}$$

$$st : \forall t \to p_{t} (C_{t}^{1} + C_{t}^{2}) + b_{t} \le w_{t}^{1} h_{t}^{1} + w_{t}^{2} h_{t}^{2} + y_{t} + b_{t-1}$$

$$0 \le h_{t}^{1} \le 1; 0 \le h_{t}^{2} \le 1$$

$$\forall t \in [0, +\infty)$$

$$\Rightarrow C_{t}^{i} = C^{i}(p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \bar{U^{2^{0}}})$$

$$\Rightarrow h_{t}^{i} = h^{i}(p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \bar{U^{2^{0}}})$$

$$\forall i = 1, 2$$

$$\Rightarrow b_{t} = b(p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \bar{U^{2^{0}}})$$

if $t \ge legal$

$$\max_{\{h_{t}^{i}, C_{t}^{i}\}_{i=1,2,t=legal,\infty}} \sum_{t=legal}^{\infty} \beta^{1} U^{1} (1 - h_{t}^{1}, C_{t}^{1}, z)$$

$$st : \sum_{t=legal}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z) \ge U^{2^{0}}$$

$$st : \forall t \rightarrow p_{t} (C_{t}^{1} + C_{t}^{2}) + b_{t} \le w_{t}^{1} h_{t}^{1} + w_{t}^{2} h_{t}^{2} + y_{t} + b_{t-1}$$

$$0 \le h_{t}^{1} \le 1; 0 \le h_{t}^{2} \le 1$$

$$\forall t \in [legal, +\infty)$$

$$\Rightarrow C_t^i = C^i(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^{\overline{legal}}})$$
$$\Rightarrow h_t^i = h^i(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^{\overline{legal}}})$$
$$\forall i = 1, 2$$
$$\Rightarrow b_t = b(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^{\overline{legal}}})$$

Theorem. Under commitment with renegotiation, the above programs are equivalent to the existence of the paretoweights μ^0 and μ^{legal} , so that spouses solve the following optimizations:

if t < legal

$$\begin{aligned} \max_{\{h_{t}^{i}, C_{t}^{i}\}_{i=1,2;t=0,\infty}} \mu^{0} \sum_{t=0}^{\infty} \beta^{1} U^{1} (1 - h^{1}_{t}, C^{1}_{t}, z) + (1 - \mu^{0}) \sum_{t=0}^{\infty} \beta^{2} U^{2} (1 - h^{2}_{t}, C^{2}_{t}, z) \\ st : \forall t \rightarrow p_{t} (C^{1}_{t} + C^{2}_{t}) + b_{t} \leq w^{1}_{t} h^{1}_{t} + w^{2}_{t} h^{2}_{t} + y_{t} + b_{t-1} \\ 0 \leq h^{1}_{t} \leq 1; 0 \leq h^{2}_{t} \leq 1 \\ \forall t \in [0, +\infty) \\ \Rightarrow C^{i}_{t} = C^{i} (p_{t}, y_{t}, w^{1}_{t}, w^{2}_{t}, b_{t-1}, \mu^{0}) \\ \Rightarrow h^{i}_{t} = h^{i} (p_{t}, y_{t}, w^{1}_{t}, w^{2}_{t}, b_{t-1}, \mu^{0}) \\ \forall i = 1, 2 \\ \Rightarrow b_{t} = b (p_{t}, y_{t}, w^{1}_{t}, w^{2}_{t}, b_{t-1}, \mu^{0}) \end{aligned}$$

if $t \ge legal$

$$\begin{aligned} \max_{\{h_{t}^{i}, C_{t}^{i}\}_{i=1,2; t=legal}} \mu^{legal} \sum_{t=legal}^{\infty} \beta^{1} U^{1} (1 - h^{1}_{t}, C_{t}^{1}, z) + (1 - \mu^{legal}) \sum_{t=legal}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z) \\ st : \forall t \to p_{t} (C_{t}^{1} + C_{t}^{2}) + b_{t} \leq w_{t}^{1} h_{t}^{1} + w_{t}^{2} h_{t}^{2} + y_{t} + b_{t-1} \\ 0 \leq h_{t}^{1} \leq 1; 0 \leq h_{t}^{2} \leq 1 \\ \forall t \in [legal, +\infty) \end{aligned}$$
$$\Rightarrow C_{t}^{i} = C^{i} (p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \mu^{legal}) \\ \Rightarrow h_{t}^{i} = h^{i} (p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \mu^{legal}) \\ \forall i = 1, 2 \\ \Rightarrow b_{t} = b(p_{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}, \mu^{legal}) \end{aligned}$$

where μ^0 is the pareto-weight associated to the wife at the time of marriage and μ^{legal} is the pareto-weight associated to the wife at the time of the renegotiation due to the unanticipated legalization of abortion.

Proof. Lagrange theorem on the existence of Lagrange multipliers.

Before the legalization of abortion, only the distribution of bargaining power at t = 0 matters: therefore, μ^0 is affected by bargaining power variables only of time t = 0 When abortion is legalized, there is an unanticipated shock to the bargaining power; spouses renegotiate their marital contract through μ^{legal} . Then, the household sticks to the optimal allocations decided at t= legal for the whole time horizon from t = legal on. After the legalization of abortion, only the distribution of bargaining power at t = legal matters: therefore μ^{legal} and consequently the optimal allocations are affected by bargaining power variables only of time t = legal. In particular, if abortion legalization actually increases female bargaining power in the household, it must be true that:

$$\mu^{legal} > \mu^0$$

Therefore, abortion legalization should have an impact on households' observed behavior since wife's pareto-weight increases.

3) No commitment

Under no commitment, every period spouses decide their current optimal allocations, depending only on the current bargaining power. The behavior of labor supply and consumption over time may possibly change at any time after the marriage, not only in response to legalization per se but also to changes in accessibility of abortion.

In fact, the household intertemporal efficient allocations of resources can be written as the solutions of the following programs:

$$\forall t \in [0, +\infty)$$

$$\max_{\{h_t^i, C_t^i\}_{i=1,2}} U^1(1-h_t^1, C_t^1, z)$$

$$st: U^{2}(1-h_{t}^{2}, C_{t}^{2}, z) \ge U^{2^{t}}$$

$$st: p_{t}(C_{t}^{1}+C_{t}^{2})+b_{t} \le w_{t}^{1}h_{t}^{1}+w_{t}^{2}h_{t}^{2}+y_{t}+b_{t-1}$$

$$0 \le h_{t}^{1} \le 1; 0 \le h_{t}^{2} \le 1$$

$$\Rightarrow C_t^i = C^i(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^t})$$

$$\Rightarrow h_t^i = h^i(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^t})$$

$$\forall i = 1, 2$$

$$\Rightarrow b_t = b(p_t, y_t, w_t^1, w_t^2, b_{t-1}, U^{2^t})$$

Theorem. Under no commitment, the above programs are equivalent to the existence of the pareto-weights μ^t , so that

spouses solve the following optimizations:

$$\forall t \in [0, +\infty)$$

$$\max_{\{h_t^i, C_t^i\}_{i=1,2}} \mu^t U^1 (1 - h_t^1, C_t^1, z) + (1 - \mu^t) U^2 (1 - h_t^2, C_t^2, z)$$

$$st : p_t (C_t^1 + C_t^2) + b_t \leq w_t^1 h_t^1 + w_t^2 h_t^2 + y_t + b_{t-1}$$

$$0 \leq h_t^1 \leq 1; 0 \leq h_t^2 \leq 1$$

$$\Rightarrow C_t^i = C^i (p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^t)$$

$$\Rightarrow h_t^i = h^i (p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^t)$$

$$\forall i = 1, 2$$

$$\Rightarrow b_t = b(p_t, y_t, w_t^1, w_t^2, b_{t-1}, \mu^t)$$

where μ^{t} is the pareto-weight associated to the wife at time t.

Proof. Lagrange theorem on the existence of Lagrange multipliers.

In every period t, only the current distribution of bargaining power in t matters; therefore, μ^t is affected only by current bargaining power variables. With no commitment, spouses are already updating their bargaining power balance and deciding their current optimal allocations period by period, independently of the realization of an unexpected shock to their bargaining power.

When abortion is legalized they take into account of the new power balance favorable to wives. Before the legalization, $\mu^{t} = \mu^{0}$ $\forall t < legal$, and at the time of the legalization, $\mu^{t} = \mu^{legal}$ for t = legal.

Moreover, the no-commitment assumption allows for further adjustments in the power balance in response to changes in the degree of legalization, e.g. in terms of accessibility of abortion. If actual accessibility of abortion changes over time affecting wife's opportunity to directly control her fertility, spouses may change behavior accordingly. In particular, if abortion legalization actually increases female bargaining power in the household, it must be true that:

$$\mu^{legal} > \mu^{legal-k} \; \forall k \in [1, legal]$$

Moreover, if actual accessibility of abortion changes over time and, by affecting wife's opportunity to directly control her fertility, further modifies her bargaining power, it also holds that:

$$\mu^{legal+k} \stackrel{>}{\underset{<}{\rightarrow}} \mu^{legal+j} \quad \forall k \ge 0 \quad \forall j > 0$$

Therefore, abortion legalization should affect household's observed behavior since wife's pareto-weight increases.

Testable predictions on labor supply

I use the concept of sharing rule, an equivalent alternative way of representing the intra-household power balance; it determines how non-labor income is divided between the two spouses in every period t. By efficiency, it is equivalent to the pareto-weight formulation.

After illustrating this equivalence, I derive the theoretical predictions that each model yields on how the change in bargaining power due to abortion legalization affects the sharing rule; subsequently, I use the impact of the sharing rule on labor supply to link the legalization to a labor supply effect. In this way, the testable predictions on labor supply will consist of ultimately testing whether the sharing rule changes due to abortion legalization.

Specifically, the share of non-labor income going to the wife is increasing in her pareto-weight $(\frac{\partial \rho}{\partial \mu} > 0)$; a change in

her share is an income effect and implies a change in the opposite direction of her labor supply. The reverse is true for the husband.

1) Full commitment

Theorem. Pareto-efficiency is equivalent to the existence of a time invariant sharing rule $\rho^0 = \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1})$ such that the wife gets the share $\rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1})$ of the household non-labor income, the husband gets the remaining $(y_t - \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}))$ and they solve the following programs:

$$\max_{\{h_t^1, C_t^1\}_{t=0,\infty}} \sum_{t=0}^{\infty} \beta^1 U^1 (1 - h_t^1, C_t^1, z)$$

$$st : \forall t \to p_t C_t^1 + b_t^1 \le \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}) + w_t^1 h_t^1 + b_{t-1}^1$$

$$\Rightarrow h_t^1 = h^1(p_t, w_t^1, b_{t-1}, \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}))$$

$$\max_{\{h_{t}^{2}, C_{t}^{2}\}_{t=0,\infty}} \sum_{t=0}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z)$$

$$st : \forall t \to p_{t} C_{t}^{2} + b_{t}^{2} \leq y_{t} - \rho(\mu^{0}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{2} h_{t}^{2} + b_{t-1}^{2}$$

$$\Rightarrow h_{t}^{2} = h^{2} (p_{t}, w_{t}^{2}, b_{t-1}, y_{t} - \rho(\mu^{0}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$
with $b_{t}^{1} + b_{t}^{2} = b_{t} \forall t$

Proof. First and Second Welfare theorems.

Since μ^0 depends only on the distribution of bargaining power at t = 0, ρ^0 is affected by bargaining power variables only of the time of marriage t = 0. The sharing rule $\rho^0 \forall t \in [0, +\infty)$ is chosen by the two spouses ex ante at the time of the marriage (t = 0): it only depends on information available at the time of the contract and it applies to every period t for the whole time horizon. Equivalently to μ^0 , we can state that the household sticks to the sharing rule ρ^0 and to the optimal allocations she committed to in t = 0, even though the legalization of abortion may represent a distribution factor. Consequently, we can state that, everything else being equal:

•
$$\mu = \mu^0 \implies \rho = \rho^0 \quad \forall t \in [0, +\infty) \implies h_{legal+k}^1 = h_{legal-j}^1 \quad \forall k \ge 0 \quad \forall j > 0$$

•
$$\mu = \mu^0 \implies \rho = \rho^0 \quad \forall t \in [0, +\infty) \implies h_{legal+k}^2 = h_{legal-j}^2 \quad \forall k \ge 0 \quad \forall j > 0$$

Ceteris paribus, husband's and wife's respective labor supply should not change over time because of the legalization, since there is no increase (decrease) in wife's (husband's) share of non-labor income.

2) Commitment with renegotiation due to unanticipated shocks to bargaining power

Theorem. Pareto-efficiency is equivalent to the existence of the time invariant sharing rules $\rho^0 = \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1})$ and $\rho^{legal} = \rho(\mu^{legal}, y_t, w_t^1, w_t^2, b_{t-1})$, such that the wife gets the share $\rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1})$ if t < legal, or $\rho(\mu^{legal}, y_t, w_t^1, w_t^2, b_{t-1})$ if t ≥ legal, of the household non-labor income, the husband gets the remaining $(y_t - \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}))$ if t < legal, or $(y_t - \rho(\mu^{legal}, y_t, w_t^1, w_t^2, b_{t-1}))$ if t ≥ legal, and they solve the following programs:

$$\max_{\{h_{t}^{1}, C_{t}^{1}\}_{t=0,\infty}} \sum_{t=0}^{\infty} \beta^{1} U^{1} (1 - h_{t}^{1}, C_{t}^{1}, z)$$

$$st : \forall t \to p_{t} C_{t}^{1} + b_{t}^{1} \le \rho(\mu^{0}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{1} h_{t}^{1} + b_{t-1}^{1}$$

$$\Rightarrow h_{t}^{1} = h^{1}(p_{t}, w_{t}^{1}, b_{t-1}, \rho(\mu^{0}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$

$$\max_{t=0}^{\infty} \sum_{t=0}^{\infty} \beta^{2} U^{2} (1 - h^{2} C^{2} z)$$

$$\max_{\{h_t^2, C_t^2\}_{t=0,\infty}} \sum_{t=0}^{\infty} \beta^2 U^2 (1 - h_t^2, C_t^2, z)$$

$$st : \forall t \to p_t C_t^2 + b_t^2 \le y_t - \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}) + w_t^2 h_t^2 + b_{t-1}^2$$

$$\Rightarrow h_t^2 = h^2 (p_t, w_t^2, b_{t-1}, y_t - \rho(\mu^0, y_t, w_t^1, w_t^2, b_{t-1}))$$

with $b_t^1 + b_t^2 = b_t \forall t$

if $t \ge legal$

$$\max_{\{h_{t}^{1}, C_{t}^{1}\}_{t=legal,\infty}} \sum_{t=legal}^{\infty} \beta^{1} U^{1}(1-h_{t}^{1}, C_{t}^{1}, z)$$

$$st : \forall t \to p_{t} C_{t}^{1} + b_{t}^{1} \leq \rho(\mu^{legal}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{1} h_{t}^{1} + b_{t-1}^{1}$$

$$\Rightarrow h_{t}^{1} = h^{1}(p_{t}, w_{t}^{1}, b_{t-1}, \rho(\mu^{legal}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$

$$\max_{\{h_{t}^{2}, C_{t}^{2}\}_{t=legal,\infty}} \sum_{t=legal}^{\infty} \beta^{2} U^{2} (1 - h_{t}^{2}, C_{t}^{2}, z)$$

$$st : \forall t \to p_{t} C_{t}^{2} + b_{t}^{2} \leq y_{t} - \rho(\mu^{legal}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{2} h_{t}^{2} + b_{t-1}^{2}$$

$$\Rightarrow h_{t}^{2} = h^{2} (p_{t}, w_{t}^{2}, b_{t-1}, y_{t} - \rho(\mu^{legal}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$
with $b_{t}^{1} + b_{t}^{2} = b_{t} \forall t$

Proof. First and Second Welfare theorems.

Before the legalization of abortion, only the distribution of bargaining power at t = 0 matters: therefore, μ^0 and consequently ρ^0 are affected by bargaining power variables only of time t = 0.

When abortion is legalized, the sharing rule $\rho^{legal} = \rho(\mu^{legal}, y_t, w_t^1, w_t^2, b_{t-1}) \forall t \in [legal, +\infty)$ is chosen by the two spouses at the time of the legalization and it only depends on information available at the time t = legal; it applies to the whole time horizon from t = legal on.

Equivalently to μ^0 , we can state that the household sticks to the sharing rule ρ^0 and to the optimal allocations she committed to in t = 0, until the unexpected shock of abortion legalization makes her renegotiate the bargaining power and optimal allocations in t = legal, using the sharing rule ρ^{legal} . The household sticks to the sharing rule ρ^{legal} and to the optimal allocations she committed to in t = legal, from t = legal on. Consequently, we can state that, everything else being equal:

•
$$\mu^{legal} > \mu^0 \Rightarrow \rho^{legal} > \rho^0 \Rightarrow h^1_{legal+k} < h^1_{legal-j} \quad \forall k \ge 0 \quad \forall j > 0$$

• $\mu^{legal} > \mu^0 \Rightarrow \rho^{legal} > \rho^0 \Rightarrow h^2_{legal+k} > h^2_{legal-j} \quad \forall k \ge 0 \quad \forall j > 0$

Ceteris paribus, the increase in the wife's share of non-labor income represents a positive income effect, making her labor supply decrease after the legalization. The prediction for husbands has the opposite sign, since their share of non-labor income consequently decreases: bargaining power is indeed a relative concept.

3) No commitment

Theorem. Pareto-efficiency is equivalent to the existence of the sharing rules $\rho^t = \rho(\mu^t, y_t, w_t^1, w_t^2, b_{t-1})$, $\forall t \in [0, +\infty)$, such that the wife gets the share $\rho(\mu^t, y_t, w_t^1, w_t^2, b_{t-1})$ of the household non-labor income, the husband gets the remaining $(y_t - \rho(\mu^t, y_t, w_t^1, w_t^2, b_{t-1}))$, and they solve the following programs:

$$\max_{\{h_{t}^{1}, C_{t}^{1}\}} U^{1}(1 - h_{t}^{1}, C_{t}^{1}, z)$$

$$st : p_{t}C_{t}^{1} + b_{t}^{1} \leq \rho(\mu^{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{1}h_{t}^{1} + b_{t-1}^{1}$$

$$\Rightarrow h_{t}^{1} = h^{1}(p_{t}, w_{t}^{1}, b_{t-1}\rho(\mu^{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$

$$\max_{\{h_{t}^{2}, C_{t}^{2}\}} U^{2}(1 - h_{t}^{2}, C_{t}^{2}, z)$$

$$st : p_{t}C_{t}^{2} + b_{t}^{2} \leq y_{t} - \rho(\mu^{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}) + w_{t}^{2}h_{t}^{2} + b_{t-1}^{2}$$

$$\Rightarrow h_{t}^{2} = h^{2}(p_{t}, w_{t}^{2}, b_{t-1}, y_{t} - \rho(\mu^{t}, y_{t}, w_{t}^{1}, w_{t}^{2}, b_{t-1}))$$
with $b_{t}^{1} + b_{t}^{2} = b_{t} \forall t$

Proof. First and Second Welfare theorems.

The sharing rule ρ^t is chosen in each period t and it reflects the current bargaining power balance μ^t .

In every period t, only the current distribution of bargaining power in t matters: consequently, ρ^t , as μ^t is, is affected by bargaining power variables only of time t. Under no commitment, the household renegotiates the bargaining power balance and optimal allocations in each t, using the sharing rule ρ^t . Ceteris paribus, spouses use $\rho^t = \rho^0$ if t < legal and, at the time of legalization $\rho^t = \rho^{legal}$ for t = legal. Consequently, we can state that, everything else being equal:

•
$$\mu^{legal+k} > \mu^{legal-j} \Rightarrow \rho^{legal+k} > \rho^{legal-j} \quad \forall k \ge 0 \quad \forall j > 0 \Rightarrow h^1_{legal+k} < h^1_{legal-j} \quad \forall k \ge 0 \quad \forall j > 0$$

•
$$\mu^{legal+k} > \mu^{legal-j} \Rightarrow \rho^{legal+k} > \rho^{legal-j} \quad \forall k \ge 0 \quad \forall j > 0 \Rightarrow h^2_{legal+k} > h^2_{legal-j} \quad \forall k \ge 0 \quad \forall j > 0$$

If actual accessibility of abortion changes over time and correspondingly affects her bargaining power, we can also state that:

•
$$\mu^{legal+k} \stackrel{>}{\underset{<}{\sim}} \mu^{legal+j} \Rightarrow \rho^{legal+k} \stackrel{>}{\underset{<}{\sim}} \rho^{legal+j} \quad \forall k \ge 0 \; \forall j > 0 \qquad \Rightarrow h^1_{legal+k} \stackrel{<}{\underset{>}{\sim}} h^1_{legal+j} \; \forall k \ge 0$$

• $\mu^{legal+k} \stackrel{>}{\underset{<}{\sim}} \mu^{legal+j} \Rightarrow \rho^{legal+k} \stackrel{>}{\underset{<}{\sim}} \rho^{legal+j} \quad \forall k \ge 0 \; \forall j > 0 \qquad \Rightarrow h^2_{legal+k} \stackrel{>}{\underset{<}{\sim}} h^2_{legal+j} \; \forall k \ge 0$

Ceteris paribus, the increase in the wife's share of non-labor income represents a positive income effect, making her labor supply decrease after the legalization. The prediction for husbands has the opposite sign, since their share of non-labor income consequently decreases: bargaining power is indeed a relative concept. Moreover, the wife's (husband's) labor supply may further decrease (increase) after the legalization if the accessibility of abortion, measured as territorial availability of abortion clinics and/or price for an abortion surgery, increases (decreases) over time.

Before turning to the data analysis, I present a summary of the prediction that each model yields on spouses' labor supply evolution over time in response to the abortion legalization. These predictions will be tested in the data. The negative effect of abortion legalization on fertile wives' labor supply and the corresponding positive effect on their husbands' labor supply seem unique to the bargaining power effect; also, they specifically concern women in their fertile age who have a spouse. In particular, the impact on married fertile female labor supply is the opposite of those predicted for all fertile women by the fertility effect and the technology effect.

summary of testable prediction of the models

Changes due to the legalization of abortion on fertile wives' and their husbands' labor supply

Before and after the legalization

husband's ΔLs wife's ∆Ls Share of non-labor income change in her share before before / after legal before / after legal before / after legal after $\forall k \ge 0 \ \forall j > 0$ $ho^{_0}$ $\rho^{0} = \rho^{0} h_{legal+k}^{1} = h_{legal-j}^{1} h_{legal+k}^{2} = h_{legal-j}^{2}$ $ho^{\scriptscriptstyle 0}$ Full Commitment $ho^{{}^{legal}}$ ho^0 $\rho^{legal} > \rho^{0} \qquad h^{1}_{legal + k} < h^{1}_{legal - j} \quad h^{2}_{legal + k} > h^{2}_{legal - j}$ Commitment with renegotiation at t = legal $\rho^{legal - j} \rho^{legal + k} \rho^{legal + k} > \rho^{legal - j} \qquad h^1_{legal + k} < h^1_{legal - j} \qquad h^2_{legal - j} > h^2_{legal - j}$ No Commitment

$$\rho^{legal+k} \stackrel{>}{\underset{<}{\sim}} \rho^{legal+j} \quad h^{1}_{legal+k} \stackrel{<}{\underset{>}{\leftarrow}} h^{1}_{legal+j} \quad h^{2}_{legal+k} \stackrel{>}{\underset{<}{\leftarrow}} h^{2}_{legal+j}$$

III. Labor supply specifications

To test these predictions, I use the following labor supply equations:

$$h_{t}^{1} = f_{o} + f_{1} \log w_{t}^{1} + f_{2} \log w_{t}^{2} + f_{4} \log w_{t}^{1} \log w_{t}^{2} + f_{3}y_{t} + f_{5}s_{t} + f_{6}z_{t} + \text{temporal and spatial fixed effects}$$

$$h_{t}^{2} = m_{o} + m_{1} \log w_{t}^{1} + m_{2} \log w_{t}^{2} + m_{4} \log w_{t}^{1} \log w_{t}^{2} + m_{3}y_{t} + m_{5}s_{t} + m_{6}z_{t} + \text{temporal and spatial fixed effects}$$

I consider preference parameters such as age and education of each spouse, race, household composition, number of children and presence of infants, which represent a relatively standard specification in the labor supply literature. I assume that the individual discount factors and household's net savings are captured by those variables for both spouses and that the constant interest rate is also captured by the fixed effects.

Moreover, since I will focus on a data set of households both in a time series and cross-sectional dimension, the variation in legalization of abortion and its accessibility will not only be across time but across states as well.

In particular, to test for the bargaining effect of abortion legalization I will use the following specifications:

$$h_t^1 = a + b dy_legal + X \beta + \varepsilon$$

$$h_t^2 = f + g dy_{legal} + X \gamma + \varepsilon^2$$

The other regressors X include: logarithm of each spouse's wage, cross term of these logarithms, age and age squared, years of school of each spouse, household non-labor income, number of children in the household, dummy variable for the presence of infants, number of family members, dummy for blacks, logarithm of state total income as a measure of state local economy and unemployment, dummy variables for divorce laws such as unilateral and community property provisions, year fixed effects and region fixed effects (North East, South, West, with dummy variables for California and New York states and their corresponding regions adjusted)¹⁴.

The dummy variable dy_legal is equal to one if abortion is legal¹⁵. In terms of the models, it captures the possible sharing rule switching from ρ^0 to ρ^{legal} , with $\rho = \rho^{legal}$ $\forall t \ge legal$, i.e. whether abortion legalization increases female bargaining power in the household and whether households renegotiate their marital contract in response to the legalization¹⁶. It is a test of the full commitment assumption.

¹⁴ These preference parameters and controls are similar to those used in the labor supply literature, in particular in Chiappori et al. (2002) where the specific issue of bargaining power effects on labor supply is analyzed.

¹⁵ In particular, it takes value of one for the five states that legalized abortion in 1970 and subsequently for all the US from the national legalization of January 1973 on (Supreme Court ruling of Roe vs. Wade).

¹⁶ In fact, the Supreme Court decision in 1973 legalized abortion unexpectedly, so that dy_legal represents an unanticipated shock. As to state legalizations in 1970, one for sure was expected (WA); however 1970 is the earliest year in my sample so that I do not exploit the time variation of these legalizations, but just the cross-sectional difference.

In this framework, testing these predictions consists of testing if dy_legal has a negative significant impact on fertile wives' labor supply and if it has a positive significant impact on their husbands' labor supply.

Alternatively, to test the bargaining effect not only of abortion legalization per se, but of possible changes in abortion accessibility, I will use the following specifications:

$$h_t^1 = a + c dy_low + d dy_high + X \beta + \varepsilon^1$$

$$h_t^2 = f + m dy_low + n dy_high + X \gamma + \varepsilon^2$$

The dummy variable dy_low (dy_high) is equal to one if abortion is legal and there is a low (high) accessibility according to state abortion rates¹⁷ yearly ranking, a good proxy for actual availability of abortion clinics and accessibility in terms of prices and services. I use a ranking threshold to classify states according to their abortion rates, year by year¹⁸. States whose abortion rate in a given year is ranked above that threshold have dy_high equal to one (and dy_low is equal to zero); those which have it below have dy_low equal to one (and dy_high is equal to zero). The classification is updated year by year, so that those dummy variables capture the actual variation in accessibility¹⁹.

By construction, the sum of these two dummies is equal to dy_legal.

Those dummy variables capture both the legalization and the possible variation in the degree of accessibility²⁰. In terms of the models,

they capture the possible sharing rule switching from ρ^0 to ρ^{legal} , possibly with $\rho^{legal+k} \stackrel{>}{<} \rho^{legal+j} \forall k \ge 0 \quad \forall j > 0$, i.e. whether

abortion legalization and its accessibility increase female bargaining power in the household and whether households renegotiate their marital contract in response to the legalization and changes in accessibility. It is a test of the full commitment assumption versus the renegotiation assumption; it is also a test of the non-commitment assumption with renegotiation according to the evolution in abortion accessibility.

In this framework, testing the prediction of the impact of the legalization per se consists of testing if dy_low and dy_high have a negative significant impact on fertile wives' labor supply and a positive significant impact on their husbands' labor supply. Then,

¹⁷ Number of abortions per thousand live birth.

¹⁸ For each year, I rank all the fifty-one states according to their abortion rates for that year. Since the categories are updated year by year, a state can be classified differently across years, also if it exhibits the same abortion rate. Given the steady increase in abortions just after the legalization, evaluating how accessible abortion is in a given state really depends on the period of time and on the other states.

¹⁹ I do not use the abortion rates per se as a proxy for accessibility, because there is no reason to expect bargaining power to vary continuously with the actual rates: bargaining power may increase if abortion legalization represents a relevant opportunity for women, as it would be the case in "high" availability states, but not just if abortion rates increase of a given amount of units. I nevertheless run the regressions with actual abortion rates as a regressor, instead of those dummy variables; its coefficients have the right sign, even though not significant.

²⁰ I also tried to use a dummy for the presence of Medicaid funding for abortion across states and years. It turned out to be not significant across specifications, possibly because it may concern only women under Medicaid.

testing the prediction on accessibility consists of testing if these dummies have a statistically different coefficient, with the coefficient on dy_high bigger in absolute value than the dy_low one $(|d| \ge |c| \text{ and } n \ge m)^{21}$.

I additionally use singles and divorced women in their fertile age as control groups. For them, the household is composed of only one decision maker, so that the collective model collapses into the unitary one, without negotiations, reservations utility nor pareto-weight. Legalization of abortion can not affect any intra-household bargaining power by definition, since they are a single-member household. Accordingly, I will use the corresponding labor supply equation:

 $h_t = f_o + f_1 \log w_t + f_3 y_t + f_4 s_t + f_5 z_t$ + temporal and spatial fixed effects

where f_4 should be zero, since single individual's labor supply does not depend on bargaining power variables.

In particular, to test for the absence of bargaining effect of abortion legalization, I will use the following specifications:

 $h_t = a + b dy_{legal} + X \beta + \varepsilon$

or alternatively:

$$h_t = a + c dy low + d dy high + X \beta + \varepsilon$$

where in X there are the same types of regressors as before but only those concerning the single member.

All the models' testable predictions imply that these coefficients b, c and d should not be significantly different from zero.

IV. Data description

I use the Panel Study of Income Dynamics (PSID) data; this data set starts in 1968 and collects annual observations, following the same families and their split-offs over time. Households are followed through time by keeping track of the "head" of the household, who is also the interviewed person. In households including a couple, the husband is automatically considered the head in almost all cases²².

 $^{^{21}}$ If the coefficients are not statistically different, the specification collapses to the previous one with dy_legal, since by construction the latter is equal to the sum of the dy_low and dy_high.

 $^{^{22}}$ I eliminate the very few households (less than 2 % of couples) where the sex of the head is female while there being a male spouse in the family.

I use waves 3 to 13, i.e. data from 1970 to 1979. I do not use 1968 and 1969 data because some variables of interest are not recorded in these first two years, and the coding of many variables is not comparable to those of subsequent years. Wave 13 is only used to assign income and labor variables to year 1979^{23} .

My sample is composed of married, singles and divorced²⁴. The PSID code for marital status classifies couples that are cohabitants as married. This makes my sample more representative of US families that have an actual spousal interaction decision process; it should also avoid a selection bias towards couples that decide to get legally married. Moreover, there is no risk to include unstable and short-term relationships in the sample, because PSID treats cohabitants as married only from the second interview wave in which they are found to be living together²⁵. To comply with the theoretical predictions on renegotiation of marital contracts, I only keep couples who got married before the legalization of abortion²⁶.

Moreover, I also split the sample according to age of wife (or of female single or divorced), to divide fertile from non-fertile women. These two groups are kept totally separate, since the same household is not allowed to have observations in both groups. There being no fertility question in the PSID, I use an age threshold below which women are considered fertile: I consider forty years old, a standard age upper bound, at least in 1970s when medical fertility technology was not yet developed²⁷.

I exclude observations for which either the husband or the wife are different from one year to the subsequent one^{28} . This means that if a new household forms, it will enter my sample only from the subsequent year on. The main concern here is that whenever there is a major change in family structure, it is not clear to which head and wife the questions on income and labor are referred to, and also which spouses are affected by the law changes analyzed in this paper.

In addition to this, I keep only couples where the wife is actually present and exclude all households that have other income receivers besides the husband and wife. Since the focus is on the bargaining power balance within spouses, I cannot keep families where one spouse is in fact absent or households where other adults not only do live together with the couple, but represent an additional source of household income and potential decision power. Consequently, the household income variables are computed net of these other income receivers' total labor and transfer income.

²³ Later years are not included in the sample because laws, state regulations, amendments and Courts' rulings on abortion issues become much more frequent, making the classification of whether abortion is legal more problematic.

²⁴ I do not include widowed and separated couples to keep a neat distinction between multiple decision makers and one decision maker households.

 $^{^{25}}$ In fact, PSID records a "self-reported" marital status variable from 1977 on, where this distinction is actually made. Comparing these two variables at the end of the 1970s, it looks like at most 1 % of the PSID sample is cohabiting, representing around 1.5 % of all couples.

²⁶ Before the national legalization and also before the state legalization for those households living in the five states that legalized abortion in 1970. I have used different measures of "married before state legalization", including or not households which moved to those states just after 1970, requiring or not that they stay in one of those states at least up to 1972, etc. Results are robust to these definitions.

²⁷ As to non fertile women, using forty years old as lower bound should retain at most 5 % of all pregnancies. At any rate, results are robust to changes in the age threshold and in particular to its increase as lower bound for non fertile.

²⁸ To check if a household that is present in the survey for more than one year is actually composed of the same husband (wife), I check if the husband (wife) at time t is at most two years older than at time t-1: if this condition is not fulfilled, I do not include the corresponding household observation in my sample for year t. This is done through the dummies "dy_check_h" and "dy_check_w" for husband and wife respectively.

My main sample consists of working individuals (in case of couples, both spouses have positive hours of work)²⁹. All individuals are aged 25 or more, to get more "stable" and homogeneous households, for which the assumption of efficiency in the intra-household decision process is more likely to be satisfied, and especially to discard full-time students, teenagers and very young married and divorced couples.

The dependent variables, annual hours of work (of husband and wife), are defined as total annual hours of work on all jobs in a given year. The wage rate is computed in the data as the average hourly earnings, defined by total labor income divided by annual hours of work. Total family income is the total amount of resources available to a household in a given year³⁰. I construct the variable non-labor income available to husband and wife as total family income minus the labor incomes of both spouses. All income, wage and labor hour variables refer to values of the calendar year prior to the interview and thus are "lagged". All monetary variables across years are also deflated by the CPI index (for urban consumption, not seasonally adjusted, June figures) and are in 1970 \$.

Children variables consist of the total number of children aged 0-17 present in the household (there being no detailed information on children's age brackets in the interview waves under consideration), and of a dummy variable "dy_age03child" denoting a household where the youngest child present is aged at most three years old.³¹

As to education, I use the number of grades completed by husband and wife. PSID started to record this variable in 1975, topcoding the years of schooling at 17. There are no missing values for this variable and since it is a variable that remains constant over time when you are no longer a student, I attributed the 1975 values back to the same households present in years 1970 to 1974^{32} . As to race, only race of the head of the household is available, so that we cannot detect how many interracial couples there are. I construct a dummy variable (dy_black) denoting a black household head. In the PSID sample in the 1970s individuals are almost all either black or white, so that one dummy should be enough to capture the racial effect on labor supply and household behavior in general.

Outliers on total family income, wage, hours worked and number of children are excluded from the sample. Specifically, an observation is considered an outlier if it belongs to the top or bottom 2 % of the sample distribution (only top 2 % for children, otherwise we would eliminate all households without children; as to hours worked, only positive amounts are considered to compute the outliers). I also drop households where the husband, or the single decision maker, has three years of school or less. I do not remove

²⁹ This may introduce a selection bias towards non working individuals. I take into account of this bias in my empirical analysis.

³⁰ All household net assets (rents, dividends, interest) are included in "taxable" income, which in turn is included in the definition of total household income. I cannot compute any savings or any income variable net of savings because in the 1970s PSID does not record any savings variable.

³¹ Children variables, as well as non-labor income and wages, are treated as exogenous variables in my main specifications. There is some evidence that they are not likely to be endogenous: for instance, using PSID data in 1970s and married women's labor supply as I do, Mroz (1987) can not reject the exogeneity of children, non-labor income and wages. At any rate, I take into account of the possible endogeneity issues in the sensitivity section.

³² This procedure drops few households, especially because my sample mainly consists of households who got married before the legalization, so that they were already present between 1970 and 1972. Also, I believe the exact years of schooling to be a better education measure than the bracket variable "education" available in PSID from 1968 for heads and, missing between 1969 and 1971, for wives.

the so-called "poverty sample". It consists of families who were poor (below twice the poverty line) in 1966: however, by 1970 and even more so by the end of the 1970s, those households may have come out of poverty and I already exclude income outliers.

Finally, to take into account of shocks to both the overall US economy and the state local labor market conditions, and of the fact that people may be different across different areas in the US, I use years fixed effects, regions³³ or alternatively state fixed effects, and the logarithm of state total income, as a proxy for state GDP. I cannot use the more specific state unemployment rate because up to the end of the 1970s it is not available; the county unemployment rate recorded in the PSID does not help either, since it is still a bracket variable and the coding changes during the sample period. I then discard the few observations from Alaska and Hawaii and abroad in the effort to maintain a homogeneous sample.

I use PSID household weights to make the sample representative of the US population and economy.

To account for different divorce laws in this decade, I follow Chiappori et al (2002), Gray (1998), Friedberg (1998) to construct three dummies, for unilateral divorce laws, community property laws and for the joint presence of these two provisions³⁴.

My estimation sample consists of around 1897 households on average (6079 observations), of which 1491 (4869) come from multipledecision-maker households and 406 (1210) have only one decision maker. In particular, 968 (3254) come from fertile married women's households, 297 (1026) from older married women's households, 212 (672) from fertile single women and 194 (538) from fertile divorced women. For single and divorced, I focus on young women to have reasonable control groups to test the effects of abortion laws on household bargaining power.

Men work on average more annual hours than women and earn a higher hourly wage. Also, husbands are on average 3 years older than wives. As to education, on average married men have more years of schooling than married women in young couples, but this is no longer true in the other groups. Black married couples are less than one half of the number of whites, while black young single and divorced women represent around 50 % of their group. On average non-labor income is higher for older married couples than for younger and for single women, but it is somewhat lower than for divorced women. Its large variance is due to the fact that younger households have on average negative assets while older households tend to have positive assets, its maximum values are in fact much higher for older couples.

³³ I consider the PSID regions North East, North Central, South, West and Alaska-Hawaii, from which I exclude California and New York states, to add specific dummies for CA and NY, to single out these two highly populated and economically relevant states.

³⁴ The majority of states inacted the unilateral divorce provision between 1970 and 1975. In fact, these two provisions may be distribution factors potentially influencing spousal bargaining within intact households. The dummy for unilateral divorce ("dy_unil") is constructed in the following way: a state in a given year is defined to have a unilateral divorce law, if spousal consent is not required to divorce and if fault grounds are not allowed not even for property settlements and alimony. Since unilateral divorce laws per se are not favorable to women, unless they are paired with favorable laws concerning property division, I also use a dummy variable for the presence of community property laws. Married women's bargaining power is likely to be stronger in community property jurisdictions when unilateral divorce is legal, or under common law and mutual-consent divorce, if household assets are disproportionately in the husband's name. Refer to the Appendix for a scheme of states and years in which these laws are present.

V. Results

V.1 Main specifications

I run my labor supply regression specifications³⁵ on fertile married women, their husbands, and on the other "control" groups such as single, divorced women in their fertile age and married women no longer fertile.

The main specification with dy_legal and region dummies outlined in Section III yields clear results.

For fertile married women the coefficient of dy_legal is negative significant; correspondingly, their husbands exhibit a positive significant coefficient, as theory predicts. For all the other groups not affected by the abortion bargaining power effect because wives are no longer fertile or because there is no spouse, dy_legal is totally not significant. Moreover, these coefficients of fertile married women, their husbands and married non-fertile women are also statistically different³⁶.

To corroborate my results, I consider four other specifications of the same labor supply equations for the same groups of households: heckman two-step to take into account of a possible selection bias for working individuals, fixed and random effects models for households effects and state instead of region fixed effects in my main regressions. It is remarkable that all these specification confirm the negative significant effect of abortion legalization for married fertile women and a corresponding positive impact for their husbands, while the legalization appears not to be significant in explaining the labor supply pattern of single and divorced fertile women and of married non-fertile women³⁷.

Looking at the magnitude of this impact of abortion legalization on bargaining power in terms of labor supply, the decrease in annual hours worked of married fertile women is of about 200 hours at 3 % significance, while for their husbands the corresponding increase is of about 80 hours at 1 % significance. These effects are sizable, corresponding to a 15 % reduction of the average hours worked by married women below forty years old; for their husbands, the percentage becomes a 3.5 % increase of their average hours worked. Conversely, the magnitude of the coefficients for singles, divorced fertile women and for non-fertile married women is negligible, sometimes even positive and definitely not significant. The magnitude of the impact of abortion legalization is large, especially if we think that in the 1970s women are steadily increasing their labor participation and labor supply, and that abortion legalization directly affects fertility in a way that should increase female labor supply. If we compare this abortion bargaining power effect to the divorce

³⁵ I use robust standard errors (Huber/White/sandwich estimator of variance), clustered by state. Even though I do not impose the panel dimension structure in the covariance matrix, I account for the independence of households' observations between states and allow for a possible independence across households within a state; also, I use years, regions or state fixed effects in all regressions. Clustering by state should be sensible in a framework where geographic, as well as legalization variations across time and space are relevant. However, clustering by household identity yields the same results.

³⁶ I can not pool singles or divorced women together with married fertile women in order to test for the statistical difference of their coefficients, because the former are single decision maker households and hence have a distinct labor supply equation.

³⁷ However, only the heckman two-step exhibits a statistical difference in coefficients. In the PSID the cross-sectional dimension is too large compared to the time dimension, especially for only one decade of data; using fixed effects models may wipe out all the variation across households; the same may happen to the state fixed effect specification, with the presence of 48 dummies in each regression. It is also true that I cannot use household weights in the fixed and random effects models in my statistical package, as instead I do in all my other specifications; the lack of statistical difference in the fixed, random and state effects may be due to the imprecise estimates.

bargaining power effect on labor supply documented in Chiappori et al. (2002), results are similar. In their study of cross-sectional differences of divorce laws across states, the mutual provision decreased female labor supply by 117 hours and increased their husbands by 22 hours on average. My magnitudes, as well as the stronger response of women with respect to men, are comparable to theirs; nevertheless, the average hours worked in my 1970s sample of young women are lower than theirs in 1989, making my 15 % and 3.5 % around the double of their effect. These figures seem sensible of we think that my sample consists of women below forty years old and their husbands: not only younger couples may be more responsive, but on average they also work fewer hours than couples up to sixty years old. Furthermore, if other common divorce provisions such as community property division and/or enforced court payments are also in place, the changes in hours become more than 10 % for women and around 3 % for men, exactly comparable to mine³⁸. It seems reasonable that abortion legalization of 1973 by the Supreme Court ruling was sudden. Those divorce laws state variations concern particular provisions, not the legalization of divorce per se. The control that women achieve on their fertility increases their decisional power on such a relevant issue for both spouses as children, and at the same time increases their monetary power because the uncertainty in their future earnings decreases (if not their expected value increases) due to their ultimate control on fertility.

Similar evidence comes from the other labor supply specification with the availability indeces dy_low and dy_high. I use the threshold top 20 % (bottom 80 %) of state abortion rate ranking. These regressions as well show that only married fertile women and their husbands are affected by the legalization. Moreover, not only their coefficients for both dy_low80 and dy_high20 are negative significant, but the coefficient for dy_high20 is always larger (in absolute value) than and statistically different from dy_low80 in the main specifications. It seems that the bargaining power effect may be enhanced by a high accessibility of abortion services.

All other variables in the regressions show parameter estimates at conventional levels, comparable to the female labor supply literature. In particular, the dummy for blacks is significant and sizably positive for young married women, whereas it is negative but mildly significant for singles and divorced young women. The dummy for infants is instead very significant and sizably negative for young married women, and it remains negative and rather significant also for the other groups of women. As to the divorce dummies, there are spurious results, also varying across specifications, which may be due to small sample size combined with the presence of many dummy variables³⁹. For instance, in the main specification, while for married fertile women the coefficients are all negative and

³⁸ Chiappori et al. (2002) also use a cross-sectional divorce laws index of various provisions, which yields smaller effects. However, my focus is on the intertemporal effect of abortion laws on labor supply, so that I compare my results to their state divorce laws variables specification.

³⁹ Two caveats apply to interpret these divorce laws pattern. First, among the category "married" there are also cohabitants, whose bargaining power balance should not be affected by divorce laws, at least not in the same way as legally married couples. Therefore, the presence of cohabitants can generate spurious results. Second, couples may envisage the possibility of a divorce when signing the marital contract, already ruling on division of property in case of divorce, so that divorce laws may not affect couples already married in the years preceding the state law change. Finally, estimating the effects of abortion legalization without the divorce dummies in my regressions, I obtain the same results, both in terms of significance and magnitude of the coefficients.

not significant, for their husbands the cross term is very negative and significant and the other two dummies are positive significant and for older married women the cross term between unilateral and community property is positive significant.

All this evidence goes in the direction predicted by the theory and suggests that the bargaining power effect of abortion legalization is relevant and sizable.

At the same time, these results represent a rejection of the unitary model, in which no distribution factor influences household behavior.

They also suggest a rejection of the full commitment model: marital contracts are imperfect and it seems like we can not model household behavior using a full commitment approach. The significant decrease of married fertile women's labor supply, and their husbands' corresponding increase due to legalization of abortion, show that couples actually respond to shocks such as abortion legalization and renegotiate their marital contract. They are also likely to update their decisions over time following the evolution of the actual accessibility of abortion.

To further test the power of my results I analyze subgroups of married fertile households with characteristics that to some extent predict their degree of response to abortion legalization. It is interesting to explore the pattern of the bargaining power effect within these ex ante dimensions.

V.2 Religion, birth control and expected children

With the help of questions asked in PSID, I identify three dimensions through which I divide my sample of married fertile women: 1) religion, i.e. catholic versus religious denominations not totally opposing abortion; 2) use of birth control, i.e. households using birth control versus households not using any; 3) whether more children are expected, i.e. households expecting more children versus households not expecting any more. The predictions within groups are the following. Catholics should exhibit a much lower (if any) bargaining power effect due to abortion legalization than the whole group of married fertile women and in particular lower than other religions with a more "flexible" view on abortion. For the Catholic Church, abortion is a capital sin and all the focus is on the rights of the yet-to be-born child; whereas some protestant denominations, such as Methodists, Episcopalian, Presbyterian and atheist in general⁴⁰, somewhat express understanding for women's choice of unwantedness⁴¹ and the views on abortion never involve an inflexible definition as sin. As to birth control, if couples already use a contraceptive method, abortion legalization does not represent such a new opportunity to control fertility. Hence, the bargaining power effect should be much milder (if any) for this group than for

⁴⁰ I could not consider Unitarians because PSID does not single them out as a distinct group.

⁴¹ For instance, the Catholic Church states that "abortion directly violates the divine commandment "you shall not kill"....it's an attack against human life" (Evangelium Vitae, 1995), while Methodists state that "they respect the sacredness of the life and well-being of the mother, for whom devastating damage may result from an unacceptable pregnancy" (United Methodists Social Justice Stands).

the whole sample or for the households not using contraception. Supposedly female bargaining power already changed with their decision of contraception, and fertility is less likely to be a conflicting issue for these households. Finally, couples not expecting children may not consider abortion legalization as a shock to their bargaining power balance because fertility choices are no longer at stake for them: it does not really represent an opportunity that can change their decisions and give women more freedom, since fertility is presumably no longer a source of conflict. Consequently, the bargaining power effect should be much milder (if any) for this group than for the whole sample or for the households still expecting more children.

Some clarifications apply. In PSID, the question on religious preference is asked every year, while those about the use of birth control and expectations of more children have been asked only in 1972. This year is before the national legalization which is my main source of variation. The answer on expected children may be endogenous to abortion in states that legalized in 1970, before the question was asked: households living in those states may be more likely to answer that they do not expect children because they are aware of the abortion legalization and its implications for fertility. Yet, I believe this dimension is still indicative; in particular, if an endogenous bias exists it should go in the opposite direction of my prediction: it should make those answering "not expecting any more" more responsive to abortion legalization because the bargaining power effect should be already in place in those states. As to religion, two comments are needed. First, the attitude of the entirety of the believers towards abortion may not necessarily reflect the official positions of their churches on this topic. However, I believe my empirical results across religious groups are interesting since their views on abortion are clearly divergent. Second, all these questions in PSID are asked to the head of the household (husband), so that we know the religion of the head of the household. General statistics on mixed marriages and a later question on religion of the wife (not used for too small sample size) show that in around ¼ of marriages spouses are of the same religion, even more so in the 1970s⁴².

The evidence on these subsamples of married fertile women confirms these predicted attitudes. Across the main specification regressions run on catholics, the coefficient of dy_legal is totally not significant and very mildly negative, while for Methodists, Episcopalian, Presbyterian and atheist in general, dy_legal is negative and very significant and even of larger magnitude than for the whole sample. As to the birth control dimension, couples using birth control exhibit a non-significant negative coefficient on dy_legal; couples not reported to use birth control show instead a very negative and significant coefficient. The prediction of lower impact of abortion legalization on couples not expecting children is also confirmed: their coefficient of dy_legal is not significant across specifications, while for couples expecting more children dy_legal has a negative significant coefficient.

These findings, together with the results on different marital status groups, represent comforting and ample evidence against the rejection of bargaining power effects of abortion legalization and of the collective model with renegotiation.

⁴² I do not present here other dimensions, for instance whether or not there is a large age or education gap between spouses, because there are mixed results, very few observations for some subgroups and issues of selection bias.

VI. Sensitivity analysis

I performed robustness checks on my empirical results across various regression specifications and dimensions, to get a confirmation of my theoretical predictions on the bargaining power effect and its implications on renegotiation. I illustrate the main ones in what follows, while an extensive discussion of alternative explanations will be performed in the next section.

Looking at the income distribution dimension, I analyze the response to abortion legalization of rich and poor married households where the wife is in her fertile age⁴³. As for the top 10 % of per capita household income, it is interesting to notice that the coefficient of dy_legal is not at all significant and mildly negative across specifications, while for the remaining "non-rich" the coefficients are very significant and of a larger magnitude than usual. This goes along with the fact that women of wealthy families can definitely afford to get safe abortions even before the legalization, so that the event of the legalization does not really affect their opportunities for fertility decision. The bottom 10 % of the per capita household income exhibits negative coefficients of very large magnitude, but not significant unless in the regression with state fixed effects. Also, the non poor subgroup of my sample exhibit negative significant coefficients but lower in magnitude than for the whole group. In principle, poor households respond more to the bargaining power effect because the abortion legalization represents an opportunity really taken advantage of by this group as these findings suggest; however, the sample size on one hand and the fact that Medicaid funding for abortion was available only up to 1976 may create this low power result⁴⁴. The comparison across the income range seems to show that legalization of abortion increased female bargaining power more for households not at the top of the income distribution.

When I split the sample according to race, both black and white married fertile households exhibit a strong negative response to abortion legalization. It is somewhat surprising that the significance of dy_legal for whites is much higher than for blacks: even though a higher number of abortions are performed on white than black women, the abortion rate has always been higher for blacks.

In the data description section I mentioned the possible endogeneity of children. I address this issue as follows. First, in my main specifications I consider children and income variables as exogenous. My motivation is primarily that in the literature on female supply there is evidence for not rejecting this exogeneity. In particular, Mroz (1987) shows that the exogeneity of children and income variables is not rejected and he focuses on married women's labor supply, using PSID data of the 1970s exactly as I do. Second, I run the reduced form regression of the labor supply equation without children variables and also without wages⁴⁵: I find the same pattern

⁴³ I will explicitly refer to the main group of married fertile women; however, as to the other groups, results keep the same pattern as in my previous analysis.

⁴⁴ I do not use the poverty sample as a poverty cutoff. In fact, in my sample of married fertile women, the average family income and non-labor income are similar to the corresponding averages of households not belonging to the poverty sample. See also Section IV. ⁴⁵ It may be too drastic to drop the wage variables as far as intra-household behavior is analyzed: each spouse's wage is a potential source of bargaining power.

of results across specifications. The coefficient of dy_legal is negative significant for married fertile women and positive significant for their husbands, the magnitudes being very similar to my main specification. Also, for all the other groups the coefficient is not at all significant. Last, I instrument the children variables, non-labor income and wages; following Chiappori et al. (2002), I use years of school squared, father's education, dummy variable for being catholic, dummy variable for being Latino, lagged wages and non-labor income, and the other suitable regressors as instruments. In this case as well I find the same pattern of results.

The empirical results appear robust as well to modifications in the restrictions of the regression specification. Further robustness checks involve changes in the age cutoff for fertility of women (42-38 years old), in the age lower bound (22-28 years old), in the education upper bound and lower bound for each spouse (16 years of schooling instead of 17, 3 or 4 as lower bound), in the stability of marriage (same marital status in both the previous and subsequent year), in the abortion rate thresholds (top 25 % or 33 %). It may make sense to increase the age lower bound for divorced, to keep a more homogeneous group: women that around 25 years old are already divorced may differ sistematically from the rest of the sample. If we decrease the age lower bound to 22, 20 years old, the same pattern of results is observed, but with lower abortion coefficients. The sample size increases considerably, especially for fertile married women and singles; however, the probability of these new observations corresponding to a student or exhibiting teenage behavior is too high in my opinion to be included in the benchmark estimations.

All these evidence across dimensions and specifications seems to confirm that the negative impact of abortion legalization on female labor supply is unique to married fertile women and that this bargaining power effect along with its prediction of renegotiation is sizable and significant.

As last step of my empirical analysis, let's turn now to discuss in depth the issue of alternative explanations to the bargaining power effect of abortion legalization. It is challenging to investigate other phenomena possibly driving such a distinctive pattern of a decrease only of married fertile women's labor supply and of a corresponding positive impact on their husbands'.

VII. Abortion and female labor supply: alternative explanations

The explanation that the bargaining power effect, as a prediction of the intertemporal collective model with renegotiation, provides for the peculiar impact of abortion legalization on married fertile women's labor supply and their husbands' is a priori not exclusive. However, the collective model per se and our data allow testing for possible alternative explanations. In general, singles, divorced⁴⁶

⁴⁶ I also run my main specifications on single and divorced men getting the same result of no significant impact of abortion and different pattern from married fertile women's labor supply.

and older married women's labor supply movements provide a simple and powerful test for explanations alternative to my bargaining power effect and its theoretical predictions. Other explanations may imply the same pattern for singles, divorced or old married women's labor supply, since they share the same gender, age level or marital status as the main group of married fertile women. Indeed, in all our data specification, the empirical evidence clearly shows that the variables capturing abortion legalization do not have any impact on singles, divorced and older married women.

My analysis consists of four points: 1) labor market shocks; 2) fertility and marriage market effects; 3) modifications in the use of contraceptives; 4) welfare effects for women.

1) Spatial variation of labor demand, gender-specific and/or age-specific, wider economic shocks, especially after the national abortion legalization in 1973 with the first oil shock, may explain the significant decrease in labor supply of young married women. I think we can discard this explanation of downturns in the labor market, mainly for two reasons. First, in all regressions I control for wages and economy changes through years and regions fixed effects and state total income; second, these supposed labor market downturns do not seem true for young singles and divorced, and for older married women; if we think that younger workers respond more to economic shocks or that sectors demanding young female labor experienced downturns, this should hold also for young singles and divorced, but this is not what I find. Also, we know that female labor supply increased much overtime (not decreased) especially for married women. If, instead, labor demand varies according to marital status, and my abortion dummies are correlated with these unobserved labor market characteristics, we should see the same pattern for all married women's labor supply both below and above the fertile age. Finally, such a distinctive increase in men's labor supply, only for those married to fertile women, is hard to reconcile with economic downturns⁴⁷. The issue of possible measurement error caused by mobility across states in which abortion is available should not seriously affect my estimates. The fact that women living in neighbor states can go to states where abortion has been legalized would cause downward bias in the coefficient of the abortion dummies, making the true bargaining power effect even larger than in my findings. Also, this problem may raise only for variation across state laws, while my main source of variation is the national legalization of 1973.

2) The impact of abortion legalization on fertility and possibly on the marriage market as the primary sources of female labor supply movements should not represent an alternative explanation. First of all, I control for children variables (number of children aged 0-17; presence of infants), as well as for the age of wives, their wages, labor supply, marital status; second, the delay or decrease in fertility that appear to be the direct effect of the legalization should lead to an increase in female labor supply (as found in the literature) not to a decrease only for married young women and to an increase for their husbands. The decrease in labor supply may indicate that they

⁴⁷ I am planning to analyze the impact of abortion legalization also on labor market participation decisions.

spend more time with their children. However, children variables are also instrumented and results do not change⁴⁸. Moreover, as to married market changes, I focus on couples who got married before the legalization, so that there should not be any such issue by definition.

3) In general, birth control and female contraceptive methods, whose usage might have spread out in the 1970s, may yield the same bargaining power effect on labor supply as abortion legalization, because they give women an opportunity to control fertility decisions. Condoms, for instance, can not be considered as a bargaining power variable favorable to women because they require male's consent, thus not providing women with the freedom to directly control fertility and in any case their usage widely spread one decade later, in the 1980s. I focus on the availability and change in use of the major female contraceptives such as IUD, diaphragm and above all the contraceptive pill. Typically, they all give women exactly the same independence of choice as abortion legalization does, at a lower economic and emotional cost. Hence, we could not disentangle the bargaining power effect of abortion legalization from the one of female contraceptive availability if all changes happened in the same period of time. It turns out that the pill became available to married women in 1960, and its diffusion was very rapid⁴⁹; as to the other two mentioned contraceptives, they were already available in the 1960s, and the major changes in their usage happened in the 1960s. As in point 2), their other possible fertility effects should move labor supply in the opposite direction than my findings, so as not to represent an alternative explanation, and in any case they would be captured by my children variables. Furthermore, it seems like there is no relevant substitution effect from female contraceptives to abortion: specifically, US data on married women and contraceptive use show a mild increase (less than 1 % per year on average) in the percentage of married women using contraception in general and the pill, IUDs and diaphragm in particular, during the 1970s.

4) An increase in welfare benefits to women may explain a possible increase in women's bargaining power and a decrease in their labor supply, since per se they represent a positive income effect. However, if welfare benefits somewhat discourage work and if their increase concerned all young women, singles and divorced would be affected too; if they concerned all married women, they would affect also women above their fertile age, as instead it is not the case. Moreover, I control for income, wages and children in all my regressions, so that welfare mechanism and eligibility, if any, should be captured by these variables. Primarily, the most important argument against this alternative explanation is that welfare by definition concerns only low-income households, whereas my results hold also when low-income are removed from my sample, with the same strong significant decrease in married fertile women's labor supply, the significant increase in their husbands' and no effect at all for singles, divorced and older married women⁵⁰. Finally, the timing and pattern of the main welfare benefits such as AFDC, EITC and mandated benefits look different from abortion legalization. For instance, AFDC real benefit per capita decreases in 1970s and it is generally not available to two-parent families (Hoynes, Moffit).

⁴⁸ I am planning to model fertility decisions in my commitment framework.

⁴⁹ "Its use was approved by FDA in 1960 and its diffusion was so rapid that by 1965, 41 % of young married women using some form of contraception were "on the pill"" (Goldin Katz (2000)).

⁵⁰ Actually, from the previous discussion on sensitivity, we know that for non-poor households results are even stronger than for the whole sample.

Mandated benefits were affected by the federal legalization in 1978 (Gruber). EITC begins in 1975 as a modest program and then expands in the 1980s (Eissa Liebman, Moffit).

All this discussion suggests that my identification strategy is plausibly capturing the claimed effect of abortion legalization on bargaining power, without an overestimation bias due to this kind of omitted variables. However, it is possible that abortion legalization is part of deep social transformations occurring in the 1970s, that may have changed the relationship and bargaining power balance between spouses; in particular, I am referring to the feminist and sexual freedom and widespread increase in women's importance in society, of which abortion legalization and its highlighted bargaining power effect represent a major outcome⁵¹.

VIII. Conclusions

In this paper I propose, model and estimate the bargaining power effect of abortion legalization, focusing on the consequences of this legalization on female power within the household. My findings suggest that this legalization positively affected female bargaining power in households where women are in their fertile age.

In the framework of an intertemporal collective model of labor supply, with possible commitment to marital contract, I analyze households that were already formed before the legalization and I test for the renegotiation of the household's initial contract in the presence of the shock to bargaining power balance represented by abortion legalization. Results seem to reject the full commitment assumption and, overall, the unitary approach to household behavior.

I consider the early abortion state laws in 1970 and the federal legalization in 1973 along with data on state abortion rates in the 1970s. I use PSID data from 1970 to 1979. I control for various family and labor market characteristics, and also for possible selection bias for working women. I find that the legalization significantly decreased labor supply of married women in their fertile age and significantly increased their husbands', while no such effect is found for older married women nor for single and divorced women or men. It seems that during the 1970s in the US there has been a relevant impact on households' decision process and a change in family allocation of resources, along with a redistribution of bargaining power between spouses and a major impact on female labor supply for those women favored by this intra-household reallocation and for their husbands. I performed many robustness tests across various regression specifications and dimensions to establish two results. First, this frequently debated public policy had an indirect impact on female bargaining power within the household, beyond the fertility, education and marriage market effects found in the literature.

⁵¹ I control for years and region fixed effects and for variables like race, education and children in all my estimates; it is also unlikely that these social changes had exactly the same time and state pattern as abortion legalization and abortion rates did. Finally, there is no reason to believe that these transformations would not impact older married women for a whole decade, nor singles and divorced young women and that they would decrease female labor supply.

Second, the pattern of abortion legalization in that decade is important to explain those family changes. Nevertheless, it is possible that abortion legalization belongs to deeper transformations in social norms occurring in the 1970s in the US. This big symbolic movement may have changed spouses' perception and the way to address women's concerns in the household.

This paper is meant to be a contribution to the understanding of the effects of abortion legalization, specifically by introducing the new dimension of bargaining power effect. The overall impact of legalization of abortion on various aspects of society may be wider than previously thought.

References

- 1. Angrist, J., Evans, W. (1996), "Schooling and labor market consequences of the 1970 state abortion reforms", NBER working paper # 5406
- 2. Becker G. (1991), "A Treatise on the Family", Harvard University Press
- Blundell, R., Pashardes Panos., Weber Guglielmo (1993), "What do we learn about Consumer Demand Patterns from Micro data?" American Economic Review, 83, 570-597
- 4. Bourguignon, F., M. Browning and P.-A. Chiappori (1995), "The Collective Approach to Household Behaviour", Working Paper 95-04, Paris: DELTA.
- Bourguignon, F., M. Browning, P.-A. Chiappori and V. Lechene (1993), "Intra-Household Allocation of Consumption: a Model and some Evidence from French Data", Annales d'Économie et de Statistique, 29, 137–156.
- Browning, M., F. Bourguignon, P.-A. Chiappori and V. Lechene (1994), "Incomes and Outcomes: A Structural Model of Intra–Household Allocation", Journal of Political Economy, 102, 1067–1096.
- Browning M. and P.-A. Chiappori (1998), "Efficient Intra-Household Allocations: a General Characterization and Empirical Tests", Econometrica, 66, 1241-1278.
- Browning M., Meghir C., (1991), "The Effects of Male and Female Labor Supply on Commodity Demands" Econometrica 59, 925-951
- 9. Chiappori, P.-A. (1988), "Rational Household Labor Supply", Econometrica, 56, 63-89.
- 10. Chiappori, P.-A. (1992), "Collective Labor Supply and Welfare", Journal of Political Economy, 100, 437-467.
- 11. Chiappori, P.-A. and Ekeland (2001),"Household Demand: a Complete Characterization", Mimeo, University of Chicago.
- 12. Chiappori, P.-A, Fortin, B. and G. Lacroix (2002), "Household Labor Supply, Sharing Rule and the Marriage Market", Journal of Political Economy, 110-1, 37-72
- 13. Coombs, L., Fernandez, D. (1978), "Husband-wife agreement about reproductive goals", Demography, 15-1, 57-73
- Friedberg, L. (1998), "Did unilateral divorce raise divorce rates? Evidence from panel data", American Economic Review, 88, 608-627
- Gray, J.S. (1998), "Divorce-law changes, household bargaining, and married women's labor supply", American Economic Review, 88, 628-642
- 16. Goldin, C., Katz, L. (2000), "Career and marriage in the age of the pill", American Economic review, 90-2, 461-465
- 17. Goldin, C., Katz, L. (2002), "The power of the pill: oral contraceptives and women's career and marriage decisions", Journal of Political Economy, 110-4, 730-770
- 18. Heckman, J. (1974), "Shadow prices, market wages, and labor supply", Econometrica, 42-4
- Heer, D., Grossbard-Shechtman A. (1981), "The impact of the female marriage squeeze and the contraceptive revolution on sex roles and the women's liberalization movements in the United States, 1960 to 1975", Journal of Marriage and the Family, 43-1, 49-65
- 20. Héritier, F. (2002), "Masculin/Féminin II", Odile Jacob
- 21. Killingsworth, M., Heckamn, J. (1986), "Female Labor Supply: a survey", Handbook of Labor Economics, vol. 1, 103-204
- 22. Korenman, S., Kaestner, R., Joyce, T. (2002), "Consequences for infants of parental disagreement in pregnancy intention", Perspectives on sexual and reproductive health, 34-4, 198-205
- 23. Levine, P., Staiger D., Kane, T., Zimmerman D. (1996), "Roe vs. Wade and American fertility", NBER working paper # 5615

- Levitt, S., Donohue, J. III (2001), "The impact of legalized abortion on crime", Quarterly journal of economics, 116-2, 379-420
- 25. Mazzocco, M. (2001), "Essays on household intertemporal behavior", Ph.D. Dissertation, University of Chicago
- 26. Merz, J.F., Jackson C.A., Klerman J.A. (1996), "A review of abortion policy: legality, Medicaid funding, and parental involvement, 1967-1994", RAND, 96-24
- 27. Mroz, M. (1987),"The sensitivity of an empirical model of married women's hours of work to economic and statistical assumptions", Econometrica 55, 765-799
- 28. Thomas, D. (1990), "Intra-Household Resource Allocation: An Inferential Approach", Journal of Human Resources, 25, 635–664.
- 29. Thomas, D., Rubalcava, L. (2000),"Family bargaining and welfare" Mimeo RAND, UCLA.
- Thomson, E., McDonald E., Bumpass L. (1990), "Fertility desires and fertility: hers, his and theirs", Demography, 27-4, 579-588
- 31. United Nations (1998), "Basic Human Rights instruments", United Nations Centre for Human Rights
- 32. Van Ness, P. (ed.) (1999), "Debating Human Rights", Routledge
- Williams, L. (1994), "Determinants of couple agreement in US fertility decisions", Family planning perspectives, 26-4, 169-173
- 34. Yuval-Davis, N., Werbner, P. (eds.) (1999), "Women, Citizenship and difference", Zed Books

National legalization

Roe vs Wade ruling by the Supreme Court

"The detriment that the State would impose upon the pregnant woman by denying this choice altogether is apparent. Specific and direct harm medically diagnosable even in early pregnancy may be involved. Maternity, or additional offspring, **may force upon the woman a distressful life and future**. **Psychological harm** may be imminent. Mental and physical health may be taxed by child care. **There is also the distress**, for all concerned, **associated with the unwanted child**, and there is the problem of bringing a child into a family already unable, psycologically and otherwise, to care for it."

Roe vs Wade, 410 US 110, 153 (1973)

Abortion legalization in the US

NATIONAL LEGALIZATION

 1973 ROE vs WADE ruling by the SUPREME COURT constitutional right of privacy to choose abortion within specified periods; public funding

EARLY STATE LEGALIZATION

- 1970 CA NY WA AK HI: broadly available within specified periods; public funding
- 1967-1970 AR CO DE FL GA KS MD NM NC OR SC VA: only under special circumstances; no public funding

Table 1 SUMMARY STATISTICS

married ≤ 40 years old

married>40 years old

	mean (std. dev.)	min	max	mean (std. dev.)	min	max
HH characteristics						
hours worked wife	1255.893 (686.4)	50	2484	1348.605 (665.57)	49	2480
hours worked husband	2228.919 (509.4)	175	3680	1968.24 (713.07)	172	3675
log of wage of wife	.887 (.542)	-0.82	2.06	.874 (.515)	-0.82	2.06
log of wage of husband	1.396 (.431)	-0.39	2.38	1.32 (.554)	-0.36	2.37
log wage_wife X log wage husband	1.288 (.949)	-1.39	4.81	1.234 (.943)	-1.23	4.08
age of husband	33.79 (5.58)	25	57	57.59 (6.91)	35	82
age of wife	31.098 (4.207)	25	40	54.97 (5.57)	42	72
# of years of school of husband	12.71 (2.73)	4	17	11.44 (3.22)	4	17
# of years of school of wife	12.39 (2.69)	0	17	11.69 (2.52)	0	17
household non-labor income	623.347 (1580.7)	-3000	12848	1704.88 (2507.98)	-1882.09	21630
labor income wife	3617.358 (2740.6)	36.19	13753.73	3843.954 (2924.28)	65.8371	14382.21
labor income husband	9605.573 (3945.83)	295	26779.14	8368.168 (4637.39)	279.1367	23891.63
# of children 0-17 in the family	1.793 (1.172)	0	5	0.218 (.67)	0	5
dummy for presence of infants	.272 (.445)	0	1	.009 (.092)	0	1
# of family members	3.81 (1.18)	2	9	2.28 (.79)	2	9
dummy for black	0.088 (.28)	0	1	.08 (.27)	0	1
bargaining power variables						
abortion						
dummy for legal abortion	.77 (.42)	0	1	.73 (.45)	0	1
dummy for top 20 abortion rate states	.19 (.39)	0	1	.24 (.43)	0	1
dummy for bottom 80 abortion rate stat	tes .58 (.49)	0	1	.48 (.50)	0	1
divorce						
dummy for unilateral divorce	.19 (.39)	0	1	.19 (.39)	0	1
dummy for community property law	.13 (.33)	0	1	.16 (.37)	0	1
unilateral X community property law	.08 (.27)	0	1	.10 (.30)	0	1
	2254 (222)					
# observations (# HH)	3254 (968)			1026 (297)		

All monetary variables are in 1970 \$

SUMMARY STATISTICS (continued)

single women \leq 40 years old

divorced women≤ 40 years old

	mean (std. dev.)	min	max	mean (std. dev.)	min	max
HH characteristics						
hours worked	1805.621 (538.2)	180	3612	1794.51 (568.78)	192	3643
log of wage	1.212 (.456)	-0.38	2.26	1.04 (.51)	-0.32	2.36
age	28.8 (4)	25	40	31.5 (4.31)	25	40
# of years of school	14.45 (2.22)	7	17	12.62 (1.86)	6	17
household non-labor income	466.5717 (1076)	-595.09	12616	1106.32 (1321.8)	0	8152.8
labor income	6527.175 (2762.735)	229.36	16099.58	5701.98 (3040.68)	238.04	16099.58
# of children 0-17 in the family	.19 (.66)	0	5	1.37 (1.07)	0	5
dummy for presence of infants	.037 (.19)	0	1	.088 (.28)	0	1
# of family members	1.2 (.68)	1	7	2.39 (1.09)	1	7
dummy for black	0.15 (.36)	0	1	.16 (.36)	0	1
bargaining power variables						
abortion						
dummy for legal abortion	.88 (.32)	0	1	.89 (.31)	0	1
dummy for top 20 abortion rate states	.35 (.46)	0	1	.28 (.45)	0	1
dummy for bottom 80 abortion rate state		0	1	.61 (.49)	0	1
<i>v</i>						
divorce						
dummy for unilateral divorce	.29 (.45)	0	1	.21 (.41)	0	1
dummy for community property law	.17 (.37)	0	1	.18 (.38)	0	1
unilateral X community property law	.15 (.35)	0	1	.13 (.33)	0	1

observations (# HH) 672 (212)

538 (194)

All monetary variables are in 1970 \$

Coefficients of dy_legal, the dummy variable for whether abortion is legal

Labor supply regression specifications (dependent variable = annual hours worked) run on the below groups of households

		Married fertile women	Married non-fertile women	Single fertile women	Divorced fertile women	Men married to fertile women
reg	coeff. robust SE P value	-228.9209 103.9693 0.033	26.255 113.5096 0.818	44.58089 132.9282 0.74	-157.0001 180.494 0.391	84.74177 29.98411 0.007
Heckman 2 step	coeff. robust SE P value	-231.0824 110.8786 0.037	82.30643 175.1893 0.638	45.22168 112.8936 0.689	-172.7425 155.007 0.265	*
FE model	coeff. SE P value	- 188.4712 76.78318 0.014	-108.9619 89.88847 0.226	13.23701 134.1192 0.921	187.5217 169.3872 0.269	74.69084 62.1967 0.23
RE model	coeff. SE P value	-188.0799 68.24426 0.006	-126.6303 83.93027 0.131	-82.22237 124.1277 0.508	80.23523 143.4389 0.576	72.5349 54.75274 0.185
reg with dy states	coeff. robust SE P value	-159.997 68.46778 0.024	-121.4914 181.3714 0.507	-91.21757 77.50126 0.248	-193.6216 187.325 0.309	84.73326 36.59229 0.025
# observations # obs. for heck. 2	step	3254 5542	1026 2418	672 851	538 628	3254

Reg is my main specification with region fixed effects

Heckman 2 step is the heckman specification (mle) to take into account of the possible selection bias.

FE model is the fixed effects specification, where the unit of observation is the household.

RE model is the random effects specification.

Reg with dy_states is the specification with state fixed effects instead of regions.

All regressions use PSID household weights and are clustered by state (apart from the FE and RE specifications).* Men that are the actual husbands of the women in the first column: maximize

a MLE function for these men's hours taking into account that their wives' wages may be missing because some wives may not be working (to be done soon)

Table 2 (continued)

Coefficients of dy_legal, the dummy variable for whether abortion is legal

Labor supply regression specifications (dependent variable = annual hours worked) run also on single and divorced men \leq 40 years old.

		Single men ≤ 40	Divorced men \leq 40
reg	coeff.	175.7288	-27.2507
	robust SE	248.3992	276.0511
	P value	0.484	0.922
Heckman	coeff.	160.4572	-26.3934
2 step	robust SE	236.8591	268.6867
	P value	0.498	0.922
FE model	coeff.	-117.65	131.173
	SE	178.8833	485.7234
	P value	0.511	0.787
RE model	coeff.	65.84294	5.379355
	SE	151.9922	320.4882
	P value	0.665	0.987
reg with dy states	coeff. robust SE P value	115.0406 283.7595 0.688	-165.534 258.6806 0.526
# observations		575	411
# obs. for heck. 2 step		601	414

Table 2 (continued)

Labor supply regression specification with region fixed effects, run on married fertile women (dependent variable = annual hours worked) **Parameter estimates**

	coefficient (std. dev.)	P value
log of wage of wife	361.8339 (106.1094)	0.001
log of wage of husband	-39.0113 (71.61748)	0.589
log wage_wife X log wage husband	-82.99519 (71.13886)	0.25
age of husband	17.14531 (51.09891)	0.739
age of wife	51.58722 (85.4188)	0.549
# of years of school of husband	-36.06613 (9.39346)	<.001
# of years of school of wife	-3.939507 (6.458533)	0.54517
household non-labor income	0348834 (.0115154)	0.004
# of children 0-17 in the family	-126.4458 (79.19902)	0.118
dummy for presence of infants	-218.1202 (37.58655)	<.001
# of family members	21.94691 (79.47724)	0.784
dummy for black	223.6212 (42.92378)	<.001
dummy for legal abortion	-228.9209 (103.9693)	0.033
dummy for unilateral divorce	-20.98487 (49.30132)	0.672
dummy for community property law	-27.90524 (114.7418)	0.809
unilateral X community property law	-131.0397 (129.2954)	0.316
intercept	236.0305 (1225.962)	0.848

observations (# HH)

3254 (968)

Labor supply regression specification with region fixed effects (dependent variable = annual hours worked)

Coefficients of dy_low80 and dy_high20, dummy variables for whether abortion is legal and an observation is in a state ranked bottom 80 % or top 20% in the yearly ranking of state abortion rates (# of abortion per 1000 live births)

	<i>Married fertile</i> women	<i>Married non-fertile</i> women	Single fertile women	Divorced fertile women	<i>Men married to fertile</i>
	dy_low80 dy_high20	dy_low80 dy_high20	dy_low80 dy_high20	dy_low80 dy_high20	dy_low80 dy_high20
coeff. robust SE P value	-283.026 -406.024 69.24833 94.27175 <.001 <.001	20.06508-3.40724116.7178170.92830.8640.984	46.87817164.9584102.7196119.08220.6510.176	-150.522-138.05203.7031289.3980.4650.636	110.5746169.301337.4883459.549540.0050.007
test dy_low80=dy_h	<i>igh20</i> p value= 0.0131	0.79	0.0931	0.9196	0.1135
# observations	3254	1026	672	538	3254

Pooled regression (with region fixed effects) to test for the statistical difference of the dy_legal coefficients between married fertile women, their husbands and married non-fertile women.

The comparison group is men married to fertile women.

pooling married: married women fertile + married women non fertile + men married to fertile women					
	\uparrow	\$	\$		
dummy variables for each group:	dy_womanXfertile	dy_womanXnon fertile	comparison group		

All regressors are interacted with dy_womanXfertile and dy_womanXnon fertile.

Therefore, the following coefficients show exactly whether the impact of abortion legalization is statistically different across those groups

			# obs.	7534
dy_w	omanXfertile	dy_womanXnon fertile		
coeff. robust SE P value	565.0953 1643.803 0.733	-1994.89 3186.643 0.534		
dy_womanXfer	tileXdy_legal	dy_womanXnon fertileXdy_legal		dy_legal
coeff. robust SE P value	- 313.663 	-58.4868 117.818 ** 0.622		84.74177 30.02624 0.007

♣ = coefficient of dy_legal for fertile women *minus* coefficient of dy_legal for men married to fertile

→ *married fertile women *statistically different* from their husbands

**non-fertile married women not statistically different from husbands of fertile women

The other regression specifications yield the same results.

Table 4 (continued)

Pooled regression (with region fixed effects) to test for the statistical difference of the dy_legal coefficients between married fertile women, their husbands and married non-fertile women.

The comparison group is married women non fertile.

pooling married: married women fertile + married women non fertile + men married to fertile women					
	\uparrow	\uparrow	\$		
dummy variables for each group:	dy_womanXfertile	comparison group	dy_manXfertile		

All regressors are interacted with dy_womanXfertile and dy_manXfertile.

Therefore, the following coefficients show exactly whether the impact of abortion legalization is statistically different across those groups

dy_womanXfertile		dy_manXfertile	# obs.	7534
coeff. robust SE P value	2559.986 2770.81 0.36	1994.891 3186.643 0.534		
dy_womanXfer		dy_manXfertileXdy_legal		dy_legal
coeff. robust SE P value	-255.176 ♦ 101.819 * 0.016	58.48679 117.818 ** 0.622		26.25499 112.2361 0.816

• = coefficient of dy_legal for fertile women *minus* coefficient of dy_legal for non fertile women

→ *married fertile women statistically different from non fertile women
 **husbands of fertile women not statistically different from non-fertile married women

Heckman 2 step specification yields the same results.

Reg dy_states, FE and RE model specifications do not exhibit statistical difference between fertile and non-fertile married women.

Sample split according to religion: totally against abortion (catholic) versus not against it for fetus' health problem and/or unwantedness (Methodist, Episcopalian, Presbyterian, non-religious in general).

Labor supply specifications with region fixed effects, heckman 2 step and regression with state fixed effects, run on married fertile women separately for each of these religion subsamples.

Coefficients of dy_legal (1 if abortion is legal).

		totally against	versus	not against
reg	coeff. robust SE P value	-38.3774 155.3451 0.806		-270.363 71.20102 < .001
Heckman 2 step	coeff. robust SE P value	-75.6486 181.6089 0.677		-317.816 75.15555 < .001
reg with dy states	coeff. robust SE P value	- 34.0743 229.9041 0.883		-130.322 95.25729 0.179
# observations # obs. for heck. 2 step		538 1018		821 1425

Sample split according to birth control: households using birth control versus households not using any. Labor supply specifications with region fixed effects, heckman 2 step and regression with state fixed effects run on married fertile women, separately for each of these birth control subsamples.

Coefficients of dy_legal (1 if abortion is legal).

		birth control	versus	no birth control
reg	coeff. robust SE P value	-132.424 91.72249 0.157		-324.359 136.0439 0.021
Heckman 2 step	coeff. robust SE P value	-125.118 100.0996 0.211		-328.503 149.7651 0.028
reg with dy states	coeff. robust SE P value	-61.326 100.662 0.546		-257.439 102.8194 0.016
# observations # obs. for heck. 2 step		1508 2663		1708 2824

Sample split according to expecting children: households who expect to have more children versus households who do not. Labor supply specifications with region fixed effects, heckman 2 step and regression with state fixed effects, run on married fertile women separately for each of these expecting children subsamples. Coefficients of dy_legal (1 if abortion is legal).

		expect children	versus	not expecting
				children
reg	coeff. robust SE P value	-365.508 160.9679 0.028		-95.3601 85.19503 0.27
Heckman 2 step	coeff. robust SE P value	-372.739 173.1763 0.031		-98.9465 90.63988 0.275
reg with dy states	coeff. robust SE P value	-354.056 131.8403 0.01		-24.677 72.20521 0.734
# observations # obs. for heck. 2 step		1258 2009		1748 3085

Endogeneity of children

Reduced form: without children variables.

Labor supply specification with region fixed effects, heckman 2 step and regression with state fixed effects, run on the various groups of households. Coefficients of dy_legal (1 if abortion is legal).

		Married fertile women	Married non-fertile women	Single fertile women	Divorced fertile women	<i>Men married to fertile women</i>
reg	coeff. robust SE P value	-218.0587 125.3994 0.089	22.75624 109.4018 0.836	49.16694 133.4684 0.715	-143.2091 202.1719 0.484	82.95802 30.90407 0.01
Heckman 2 step	coeff. robust SE P value	-122.4078 92.25219 0.185	74.86539 185.1588 0.686	58.99938 108.8397 0.588	-165.8361 169.5992 0.328	*
reg with dy states	coeff. robust SE P value	-137.95 82.80495 0.103	-124.1756 183.4485 0.502	-87.17336 84.70196 0.311	-188.7488 190.1463 0.328	82.92927 38.36995 0.036
# observations # obs. for heck. 2	step	3254 5542	1026 2418	672 851	538 628	3254

Endogeneity of children and wages

Reduced form: without children variables and without wage variables.

Labor supply specification with region fixed effects, heckman 2 step and regression with state fixed effects, run on the various groups of households. Coefficients of dy_legal (1 if abortion is legal).

		Married fertile women	Married non-fertile women	Single fertile women	Divorced fertile women	<i>Men married to fertile women</i>
reg	coeff. robust SE P value	-217.449 133.4824 0.11	-16.97248 121.4101 0.89	55.80685 121.7724 0.65	-143.3301 203.7939 0.487	54.10276 30.82676 0.086
Heckman 2 step	coeff. robust SE P value	-217.797 133.2733 0.102	47.35859 207.2481 0.819	51.22016 115.2914 0.657	-164.9558 190.4857 0.387	*
reg with dy states	coeff. robust SE P value	-131.44 90.59801 0.154	-151.5257 185.6277 0.419	-98.70588 110.4946 0.379	-185.3996 192.1337 0.342	58.05851 35.8832 0.113
# observations # obs. for heck.	2 step	3254 5542	1026 2418	672 851	538 628	3254

Endogeneity of children, non-labor income and wages

Instrumental variables

instrumented: children variables, # of family members, non-labor income, wages instruments: squared years of school, father's education, dummy variable for catholic, dummy variable for latinos, lagged wages, lagged non-labor income, usual regressors.

Labor supply instrumental variable specifications with region fixed effects and with state fixed effects, run on the various groups of households. Coefficients of dy_legal (1 if abortion is legal).

		Married fertile women	Married non-fertile women	Single fertile women	Divorced fertile women	Men married to fertile women
reg	coeff. SE P value	-251.2272 127.1843 0.055	- 70.11441 174.7398 0.69	8.779538 231.994 0.97	63.72698 1159.551 0.957	65.89961 74.79564 0.383
reg with dy states	coeff. SE P value	-255.1584 210.8705 0.233	-238.4495 307.8587 0.443	-102.312 149.4029 0.499	3140.038 31884.83 0.922	85.35779 89.53028 0.346
# observations		2789	929	644	508	2789

Sample split according to race: black versus white.

Labor supply specification with region fixed effects, run on married fertile women separately for each of these race subsamples.

Coefficients of dy_legal (1 if abortion is legal).

	RACE	E black	versus	white
reg	coeff. robust SE P value	-219.971 270.002 0.422		-231.997 90.06161 0.013
Heckman 2 step	coeff. robust SE P value	-474.748 85.18445 < .001		-241.484 93.07167 0.009
reg with dy states	coeff. robust SE P value	-258.101 313.1801 0.417		-139.144 52.51451 0.011
# observations # obs. for heck. 2 step		918 1358		2212 3987

Sample split according to income: top 10 % of the per capita household income distribution versus the remaining bottom 90 %. Labor supply specification with region fixed effects, run on married fertile women separately for each of these income subsamples. Coefficients of dy_legal (1 if abortion is legal).

	RICH	rich	versus	not rich
reg	coeff. robust SE P value	-28.1744 158.6374 0.86		-263.207 100.5283 0.012
Heckman 2 step	coeff. robust SE P value	-85.1317 103.7555 0.412		-244.891 117.7303 0.038
reg with dy states	coeff. robust SE P value	125.6645 180.3523 0.491		-211.249 80.10892 0.012
# observations # obs. for heck. 2 step		234 377		3020 5165

Sample split according to income: bottom 10 % of the per capita household income distribution versus the remaining top 90 %. Labor supply specification with region fixed effects, run on married fertile women separately for each of these income subsamples. Coefficients of dy_legal (1 if abortion is legal).

	POOR	poor	versus	not poor
reg	coeff. robust SE P value	-213.612 162.7115 0.199		-221.87 102.6152 0.036
Heckman 2 step	coeff. robust SE P value	-1178.76 93787.22 0.99		-223.899 112.6973 0.047
reg with dy states	coeff. robust SE P value	-392.003 195.41 0.053		-148.801 64.94011 0.027
# observations # obs. for heck. 2 step		412 602		2842 4940

Divorce laws in the US unilateral consent and community property

• 1970-1975 AZ CA CO IN IA ME MN MT NE NV NM OK OR WA unilateral w/o fault grounds and lengthy separation

AZ CA ID LA NV NM TX WA community property already in place

- dy_unil = 1 if there is the unilateral provision
- dy_comprop = 1 if there is the community property provision
- unil_X_comprop = 1 if there are both these provisions
- unilateral per se not favorable to women

while

unilateral combined with community property should increase female bargaining power

Gray (1998); Friedberg (1998); Chiappori et al. (2002)

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI Fondazione Eni Enrico Mattei Working Paper Series Our working papers are available on the Internet at the following address

Our working papers are available on the Internet at the following addresses:

http://www.feem.it/web/activ/_wp.html

http://papers.ssrn.com

SUST	1.2002	K. TANO, M.D. FAMINOW, M. KAMUANGA and B. SWALLOW: Using Conjoint Analysis to Estimate Farmers'
		Preferences for Cattle Traits in West Africa
ETA	2.2002	Efrem CASTELNUOVO and Paolo SURICO: What Does Monetary Policy Reveal about Central Bank's
		Preferences?
WAT	3.2002	Duncan KNOWLER and Edward BARBIER: The Economics of a "Mixed Blessing" Effect: A Case Study of the
	5.2002	Black Sea
CLIM	4.2002	
		Andreas LÖSCHEL: Technological Change in Economic Models of Environmental Policy: A Survey
VOL	5.2002	Carlo CARRARO and Carmen MARCHIORI: Stable Coalitions
CLIM	6.2002	Marzio GALEOTTI, Alessandro LANZA and Matteo MANERA: Rockets and Feathers Revisited: An International
		Comparison on European Gasoline Markets
ETA	7.2002	Effrosyni DIAMANTOUDI and Effichios S. SARTZETAKIS: Stable International Environmental Agreements: An
		Analytical Approach
KNOW	8.2002	Alain DESDOIGTS: Neoclassical Convergence Versus Technological Catch-up: A Contribution for Reaching a
		Consensus
NRM	9.2002	Giuseppe DI VITA: Renewable Resources and Waste Recycling
KNOW	10.2002	Giorgio BRUNELLO: Is Training More Frequent when Wage Compression is Higher? Evidence from 11
	10.2002	European Countries
ETA	11.2002	Mordecai KURZ, Hehui JIN and Maurizio MOTOLESE: Endogenous Fluctuations and the Role of Monetary
LIA	11.2002	Policy
KNOW	12 2002	
KNOW	12.2002	Rever GERLAGH and Marjan W. HOFKES: Escaping Lock-in: The Scope for a Transition towards Sustainable
1014	12 2002	Growth?
NRM	13.2002	Michele MORETTO and Paolo ROSATO: The Use of Common Property Resources: A Dynamic Model
CLIM	14.2002	Philippe QUIRION: Macroeconomic Effects of an Energy Saving Policy in the Public Sector
CLIM	15.2002	Roberto ROSON: Dynamic and Distributional Effects of Environmental Revenue Recycling Schemes:
		Simulations with a General Equilibrium Model of the Italian Economy
CLIM	16.2002	Francesco RICCI (1): Environmental Policy Growth when Inputs are Differentiated in Pollution Intensity
ETA	17.2002	Alberto PETRUCCI: Devaluation (Levels versus Rates) and Balance of Payments in a Cash-in-Advance
		Economy
Coalition	18.2002	László Á. KÓCZY (liv): The Core in the Presence of Externalities
Theory		Laszlo A. KOCZI (IIV). The Core in the Fresence of Externations
Network		
Coalition	19.2002	
Theory		Steven J. BRAMS, Michael A. JONES and D. Marc KILGOUR (liv): Single-Peakedness and Disconnected
Network		Coalitions
Coalition	20.2002	
Theory	20.2002	Guillaume HAERINGER (liv): On the Stability of Cooperation Structures
Network		
	21 2002	Energy CAVALLADO and Luis CIDAOLO. Essentia and Environmental Sustainability A Demonia Ammenda
NRM	21.2002	Fausto CAVALLARO and Luigi CIRAOLO: Economic and Environmental Sustainability: A Dynamic Approach
ar n (in Insular Systems
CLIM	22.2002	Barbara BUCHNER, Carlo CARRARO, Igor CERSOSIMO and Carmen MARCHIORI: Back to Kyoto? US
~ ~ ~		Participation and the Linkage between R&D and Climate Cooperation
CLIM	23.2002	Andreas LÖSCHEL and ZhongXIANG ZHANG: The Economic and Environmental Implications of the US
		Repudiation of the Kyoto Protocol and the Subsequent Deals in Bonn and Marrakech
ETA	24.2002	Marzio GALEOTTI, Louis J. MACCINI and Fabio SCHIANTARELLI: Inventories, Employment and Hours
CLIM	25.2002	Hannes EGLI: Are Cross-Country Studies of the Environmental Kuznets Curve Misleading? New Evidence from
		Time Series Data for Germany
ETA	26.2002	Adam B. JAFFE, Richard G. NEWELL and Robert N. STAVINS: Environmental Policy and Technological
		Change
SUST	27.2002	Joseph C. COOPER and Giovanni SIGNORELLO: Farmer Premiums for the Voluntary Adoption of
		Conservation Plans
SUST	28.2002	<u>The ANSEA Network:</u> Towards An Analytical Strategic Environmental Assessment
KNOW	29.2002	Paolo SURICO: Geographic Concentration and Increasing Returns: a Survey of Evidence
ETA	30.2002	<i>Robert N. STAVINS:</i> Lessons from the American Experiment with Market-Based Environmental Policies
	20.2002	model visit start, and, hebbond nom mer mereen haperment with murket bubed high onderful i unered

NRM	31.2002	Carlo GIUPPONI and Paolo ROSATO: Multi-Criteria Analysis and Decision-Support for Water Management at
INIXIVI	51.2002	the Catchment Scale: An Application to Diffuse Pollution Control in the Venice Lagoon
NRM	32.2002	Robert N. STAVINS: National Environmental Policy During the Clinton Years
KNOW	33.2002	A. SOUBEYRAN and H. STAHN : Do Investments in Specialized Knowledge Lead to Composite Good
		Industries?
KNOW	34.2002	G. BRUNELLO, M.L. PARISI and Daniela SONEDDA: Labor Taxes, Wage Setting and the Relative Wage
CLIM	35.2002	Effect C. BOEMARE and P. QUIRION (Iv): Implementing Greenhouse Gas Trading in Europe: Lessons from
CLIM	55.2002	Economic Theory and International Experiences
CLIM	36.2002	<i>T.TIETENBERG</i> (lv): The Tradable Permits Approach to Protecting the Commons: What Have We Learned?
CLIM	37.2002	K. REHDANZ and R.J.S. TOL (lv): On National and International Trade in Greenhouse Gas Emission Permits
CLIM	38.2002	C. FISCHER (lv): Multinational Taxation and International Emissions Trading
SUST	39.2002	G. SIGNORELLO and G. PAPPALARDO: Farm Animal Biodiversity Conservation Activities in Europe under
		the Framework of Agenda 2000
NRM	40.2002	S.M. CAVANAGH, W. M. HANEMANN and R. N. STAVINS: Muffled Price Signals: Household Water Demand
NRM	41.2002	<u>under Increasing-Block Prices</u> A. J. PLANTINGA, R. N. LUBOWSKI and R. N. STAVINS: <u>The Effects of Potential Land Development on</u>
	41.2002	Agricultural Land Prices
CLIM	42.2002	C. OHL (lvi): Inducing Environmental Co-operation by the Design of Emission Permits
CLIM	43.2002	J. EYCKMANS, D. VAN REGEMORTER and V. VAN STEENBERGHE (lvi): Is Kyoto Fatally Flawed? An
		Analysis with MacGEM
CLIM	44.2002	A. ANTOCI and S. BORGHESI (Ivi): Working Too Much in a Polluted World: A North-South Evolutionary
		Model
ETA	45.2002	P. G. FREDRIKSSON, Johan A. LIST and Daniel MILLIMET (lvi): Chasing the Smokestack: Strategic
ETA	46.2002	Policymaking with Multiple Instruments
SUST		<i>Z. YU</i> (lvi): <u>A Theory of Strategic Vertical DFI and the Missing Pollution-Haven Effect</u> <i>Y. H. FARZIN</i> : Can an Exhaustible Resource Economy Be Sustainable?
SUST	47.2002 48.2002	Y. H. FARZIN: Sustainability and Hamiltonian Value
KNOW	48.2002 49.2002	C. PIGA and M. VIVARELLI: Cooperation in R&D and Sample Selection
	49.2002 50.2002	M. SERTEL and A. SLINKO (liv): Ranking Committees, Words or Multisets
Coalition	30.2002	M. SERTEL and A. SLIVKO (IIV). Kanking Committees, words of Munisels
Theory Network		
Coalition	51.2002	Sergio CURRARINI (liv): Stable Organizations with Externalities
Theory	51.2002	Sergio Contonina (IIV). <u>Statote organizations with Externatives</u>
Theory		
Network	52,2002	<i>Robert N. STAVINS</i> : Experience with Market-Based Policy Instruments
Network ETA	52.2002 53.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and
Network	52.2002 53.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and
Network ETA		
Network ETA ETA	53.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation
Network ETA ETA CLIM ETA	53.2002 54.2002 55.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies
Network ETA ETA CLIM ETA SUST	53.2002 54.2002 55.2002 56.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs
Network ETA ETA CLIM ETA	53.2002 54.2002 55.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of
Network ETA ETA CLIM ETA SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests
Network ETA ETA CLIM ETA SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002 58.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy
Network ETA ETA CLIM ETA SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions
Network ETA ETA CLIM ETA SUST SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation?
Network ETA ETA CLIM ETA SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002 58.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together
Network ETA ETA CLIM ETA SUST SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary
Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union
Network ETA ETA CLIM ETA SUST SUST SUST	53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic
Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity
Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary <u>Union</u> Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on <u>Tobin's Q</u> Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary <u>Union</u> Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): <u>Towards a Better Climate Treaty</u> Richard G. NEWELL and Robert N. STAVINS: <u>Cost Heterogeneity and the Potential Savings from Market- Based Policies</u> Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): <u>Reorganisation of Environmental Policy in Russia: The Decade of</u> Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): <u>Policy in Transition: New Framework for Russia's Climate Policy</u> Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): <u>How Can Economics in Transition Pursue Emissions</u> <u>Trading or Joint Implementation?</u> Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: <u>Staving Together</u> or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary <u>Union</u> <i>Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER</i> : <u>Interpreting Sustainability in Economic</u> Terms: Dynamic Efficiency Plus Intergenerational Equity <i>Carlo CAPUANO</i> : <u>Demand Growth, Entry and Collusion Sustainability</u> <i>Federico MUNARI and Raffaele ORIANI</i> : <u>Privatization and R&D Performance: An Empirical Analysis Based on</u> <u>Tobin's Q</u> <i>Federico MUNARI and Maurizio SOBRERO</i> : <u>The Effects of Privatization on R&D Investments and Patent</u> <u>Productivity</u> <i>Orley ASHENFELTER and Michael GREENSTONE</i> : <u>Using Mandated Speed Limits to Measure the Value of a</u> <u>Statistical Life</u>
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 66.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 66.2002 67.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (Iii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (Iii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K.
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 64.2002 65.2002 66.2002 66.2002 67.2002 68.2002 	 C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules; the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 64.2002 65.2002 66.2002 66.2002 67.2002 68.2002 69.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (vii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty. Taxes or Tradable Permits?
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 64.2002 65.2002 66.2002 66.2002 67.2002 68.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (vii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QURION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? Anna ALBERINI, Patrizia RIGANTI and Alberto LONGO: Can People Value the Aesthetic and Use Services of
Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM	 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 64.2002 65.2002 66.2002 66.2002 67.2002 68.2002 69.2002 	C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (vii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty. Taxes or Tradable Permits?

NRM	72.2002	Philippe BONTEMS and Pascal FAVARD: Input Use and Capacity Constraint under Uncertainty: The Case of
	72 2002	Irrigation
PRIV	73.2002	Mohammed OMRAN: The Performance of State-Owned Enterprises and Newly Privatized Firms: Empirical
PRIV	74.2002	Evidence from Egypt Mike BURKART, Fausto PANUNZI and Andrei SHLEIFER: <u>Family Firms</u>
PRIV	75.2002	<i>Emmanuelle AURIOL, Pierre M. PICARD:</i> Privatizations in Developing Countries and the Government Budget
I KI V	13.2002	Constraint
PRIV	76.2002	Nichole M. CASTATER: Privatization as a Means to Societal Transformation: An Empirical Study of
		Privatization in Central and Eastern Europe and the Former Soviet Union
PRIV	77.2002	Christoph LÜLSFESMANN: Benevolent Government, Managerial Incentives, and the Virtues of Privatization
PRIV	78.2002	Kate BISHOP, Igor FILATOTCHEV and Tomasz MICKIEWICZ: Endogenous Ownership Structure: Factors
		Affecting the Post-Privatisation Equity in Largest Hungarian Firms
PRIV	79.2002	Theodora WELCH and Rick MOLZ: How Does Trade Sale Privatization Work?
		Evidence from the Fixed-Line Telecommunications Sector in Developing Economies
PRIV	80.2002	Alberto R. PETRUCCI: Government Debt, Agent Heterogeneity and Wealth Displacement in a Small Open
CL D.(01 2002	Economy
CLIM	81.2002	Timothy SWANSON and Robin MASON (lvi): The Impact of International Environmental Agreements: The Case
PRIV	82.2002	of the Montreal Protocol George R.G. CLARKE and Lixin Colin XU: Privatization, Competition and Corruption: How Characteristics of
IKIV	82.2002	Bribe Takers and Payers Affect Bribe Payments to Utilities
PRIV	83.2002	Massimo FLORIO and Katiuscia MANZONI: The Abnormal Returns of UK Privatisations: From Underpricing
	00.2002	to Outperformance
NRM	84.2002	Nelson LOURENÇO, Carlos RUSSO MACHADO, Maria do ROSÁRIO JORGE and Luís RODRIGUES: An
		Integrated Approach to Understand Territory Dynamics. The Coastal Alentejo (Portugal)
CLIM	85.2002	Peter ZAPFEL and Matti VAINIO (lv): Pathways to European Greenhouse Gas Emissions Trading History and
		Misconceptions
CLIM	86.2002	Pierre COURTOIS: Influence Processes in Climate Change Negotiations: Modelling the Rounds
ETA	87.2002	Vito FRAGNELLI and Maria Erminia MARINA (Iviii): Environmental Pollution Risk and Insurance
ETA	88.2002	Laurent FRANCKX (lviii): Environmental Enforcement with Endogenous Ambient Monitoring
ETA	89.2002	<i>Timo GOESCHL and Timothy M. SWANSON</i> (lviii): Lost Horizons. The noncooperative management of an evolutionary biological system.
ETA	90.2002	Hans KEIDING (lviii): Environmental Effects of Consumption: An Approach Using DEA and Cost Sharing
ETA	91.2002	Wietze LISE (Iviii): <u>A Game Model of People's Participation in Forest Management in Northern India</u>
CLIM	92.2002	Jens HORBACH: Structural Change and Environmental Kuznets Curves
ETA	93.2002	Martin P. GROSSKOPF: Towards a More Appropriate Method for Determining the Optimal Scale of Production
		Units
VOL	94.2002	Scott BARRETT and Robert STAVINS: Increasing Participation and Compliance in International Climate Change
		Agreements
CLIM	95.2002	Banu BAYRAMOGLU LISE and Wietze LISE: Climate Change, Environmental NGOs and Public Awareness in
CL D (06 0000	the Netherlands: Perceptions and Reality
CLIM	96.2002	Matthieu GLACHANT: The Political Economy of Emission Tax Design in Environmental Policy
KNOW	97.2002	Kenn ARIGA and Giorgio BRUNELLO: Are the More Educated Receiving More Training? Evidence from
ETA	98.2002	<u>Thailand</u> Gianfranco FORTE and Matteo MANERA: Forecasting Volatility in European Stock Markets with Non-linear
	70.2002	GARCH Models
ETA	99.2002	Geoffrey HEAL: Bundling Biodiversity
ETA	100.2002	Geoffrey HEAL, Brian WALKER, Simon LEVIN, Kenneth ARROW, Partha DASGUPTA, Gretchen DAILY, Paul
		EHRLICH, Karl-Goran MALER, Nils KAUTSKY, Jane LUBCHENCO, Steve SCHNEIDER and David
		STARRETT: Genetic Diversity and Interdependent Crop Choices in Agriculture
ETA	101.2002	Geoffrey HEAL: Biodiversity and Globalization
VOL	102.2002	Andreas LANGE: Heterogeneous International Agreements – If per capita emission levels matter
ETA	103.2002	Pierre-André JOUVET and Walid OUESLATI: Tax Reform and Public Spending Trade-offs in an Endogenous
ET A	104 2002	Growth Model with Environmental Externality
ETA	104.2002	Anna BOTTASSO and Alessandro SEMBENELLI: Does Ownership Affect Firms' Efficiency? Panel Data Evidence on Italy
PRIV	105.2002	Bernardo BORTOLOTTI, Frank DE JONG, Giovanna NICODANO and Ibolya SCHINDELE: Privatization and
I KI V	105.2002	Stock Market Liquidity
ETA	106.2002	Haruo IMAI and Mayumi HORIE (lviii): Pre-Negotiation for an International Emission Reduction Game
PRIV	107.2002	Sudeshna GHOSH BANERJEE and Michael C. MUNGER: Move to Markets? An Empirical Analysis of
		Privatisation in Developing Countries
PRIV	108.2002	Guillaume GIRMENS and Michel GUILLARD: Privatization and Investment: Crowding-Out Effect vs Financial
		Diversification
PRIV	109.2002	Alberto CHONG and Florencio LÓPEZ-DE-SILANES: Privatization and Labor Force Restructuring Around the
	110 2002	World
PRIV	110.2002	Nandini GUPTA: <u>Partial Privatization and Firm Performance</u> François DEGEORGE, Dirk JENTER, Alberto MOEL and Peter TUFANO: <u>Selling Company Shares to</u>
PRIV	111.2002	Reluctant Employees: France Telecom's Experience
		Nordount Employees, France Ferebolin 5 Experience

PRIV	112.2002	Isaac OTCHERE: Intra-Industry Effects of Privatization Announcements: Evidence from Developed and
	112 2002	Developing Countries
PRIV	113.2002	Yannis KATSOULAKOS and Elissavet LIKOYANNI: Fiscal and Other Macroeconomic Effects of Privatization
PRIV	114.2002	Guillaume GIRMENS: Privatization, International Asset Trade and Financial Markets
PRIV	115.2002	D. Teja FLOTHO: <u>A Note on Consumption Correlations and European Financial Integration</u>
PRIV	116.2002	Ibolya SCHINDELE and Enrico C. PEROTTI: Pricing Initial Public Offerings in Premature Capital Markets: The Case of Hungary
PRIV	1.2003	Gabriella CHIESA and Giovanna NICODANO: Privatization and Financial Market Development: Theoretical Issues
PRIV	2.2003	Ibolya SCHINDELE: Theory of Privatization in Eastern Europe: Literature Review
PRIV	3.2003	Wietze LISE, Claudia KEMFERT and Richard S.J. TOL: <u>Strategic Action in the Liberalised German Electricity</u> Market
CLIM	4.2003	Laura MARSILIANI and Thomas I. RENSTRÖM: Environmental Policy and Capital Movements: The Role of Government Commitment
KNOW	5.2003	Rever GERLAGH: Induced Technological Change under Technological Competition
ETA	6.2003	<i>Efrem CASTELNUOVO</i> : Squeezing the Interest Rate Smoothing Weight with a Hybrid Expectations Model
SIEV	7.2003	Anna ALBERINI, Alberto LONGO, Stefania TONIN, Francesco TROMBETTA and Margherita TURVANI: The
SIL V	7.2005	Role of Liability, Regulation and Economic Incentives in Brownfield Remediation and Redevelopment: Evidence from Surveys of Developers
NDM	8.2003	Elissaios PAPYRAKIS and Reyer GERLAGH: <u>Natural Resources: A Blessing or a Curse?</u>
NRM		A. CAPARRÓS, JC. PEREAU and T. TAZDAÏT: North-South Climate Change Negotiations: a Sequential Game
CLIM	9.2003	
KNOW	10 2002	with Asymmetric Information
	10.2003	Giorgio BRUNELLO and Daniele CHECCHI: School Quality and Family Background in Italy
CLIM	11.2003	Efrem CASTELNUOVO and Marzio GALEOTTI: Learning By Doing vs Learning By Researching in a Model of
KNOW	12 2002	<u>Climate Change Policy Analysis</u>
KNOW	12.2003	<u>Carole MAIGNAN, Gianmarco OTTAVIANO and Dino PINELLI (eds.)</u> : Economic Growth, Innovation, Cultural Diversity: What are we all talking about? A critical survey of the state-of-the-art
KNOW	13.2003	Carole MAIGNAN, Gianmarco OTTAVIANO, Dino PINELLI and Francesco RULLANI (lvix): <u>Bio-Ecological</u> Diversity vs. Socio-Economic Diversity. A Comparison of Existing Measures
KNOW	14.2003	Maddy JANSSENS and Chris STEYAERT (lvix): Theories of Diversity within Organisation Studies: Debates and Future Trajectories
KNOW	15.2003	Tuzin BAYCAN LEVENT, Enno MASUREL and Peter NIJKAMP (lvix): Diversity in Entrepreneurship: Ethnic and Female Roles in Urban Economic Life
KNOW	16.2003	Alexandra BITUSIKOVA (lvix): Post-Communist City on its Way from Grey to Colourful: The Case Study from
KNOW	17.2003	Slovakia Billy E. VAUGHN and Katarina MLEKOV (lvix): A Stage Model of Developing an Inclusive Community
KNOW	18.2003	Selma van LONDEN and Arie de RUIJTER (lvix): Managing Diversity in a Glocalizing World
Coalition	19.2003	Sergio CURRARINI: On the Stability of Hierarchies in Games with Externalities
Theory	17.2005	So give contention. On the Stating of Includences in Games with Externatives
Network		
PRIV	20.2003	Giacomo CALZOLARI and Alessandro PAVAN (lvx): Monopoly with Resale
PRIV	21.2003	Claudio MEZZETTI (lvx): Auction Design with Interdependent Valuations: The Generalized Revelation
1101	21.2005	Principle, Efficiency, Full Surplus Extraction and Information Acquisition
PRIV	22.2003	Marco LiCalzi and Alessandro PAVAN (lvx): <u>Tilting the Supply Schedule to Enhance Competition in Uniform-</u> Price Auctions
PRIV	23.2003	David ETTINGER (lvx): Bidding among Friends and Enemies
PRIV	24.2003	Hannu VARTIAINEN (Ivx): Auction Design without Commitment
PRIV	25.2003	Matti KELOHARJU, Kjell G. NYBORG and Kristian RYDQVIST (lvx): Strategic Behavior and Underpricing in
I KI V	25.2005	Uniform Price Auctions: Evidence from Finnish Treasury Auctions
PRIV	26.2003	Christine A. PARLOUR and Uday RAJAN (lvx): Rationing in IPOs
PRIV	27.2003	<i>Kjell G. NYBORG and Ilya A. STREBULAEV</i> (lvx): <u>Multiple Unit Auctions and Short Squeezes</u>
PRIV	28.2003	Anders LUNANDER and Jan-Eric NILSSON (lvx): Taking the Lab to the Field: Experimental Tests of
		Alternative Mechanisms to Procure Multiple Contracts
PRIV	29.2003	TangaMcDANIEL and Karsten NEUHOFF (lvx): Use of Long-term Auctions for Network Investment
PRIV	30.2003	Emiel MAASLAND and Sander ONDERSTAL (lvx): Auctions with Financial Externalities
ETA	31.2003	Michael FINUS and Bianca RUNDSHAGEN: <u>A Non-cooperative Foundation of Core-Stability in Positive</u>
		Externality NTU-Coalition Games
KNOW	32.2003	Michele MORETTO: Competition and Irreversible Investments under Uncertainty
PRIV	33.2003	Philippe QUIRION: Relative Quotas: Correct Answer to Uncertainty or Case of Regulatory Capture?
KNOW	34.2003	Giuseppe MEDA, Claudio PIGA and Donald SIEGEL: On the Relationship between R&D and Productivity: A
ETA	35.2003	<u>Treatment Effect Analysis</u> Alessandra DEL BOCA, Marzio GALEOTTI and Paola ROTA: <u>Non-convexities in the Adjustment of Different</u> <u>Capital Inputs: A Firm-level Investigation</u>
		Cupius inputs. 23.1 min-itevel invosugation

GG	36.2003	Matthieu GLACHANT: Voluntary Agreements under Endogenous Legislative Threats
PRIV	37.2003	Narjess BOUBAKRI, Jean-Claude COSSET and Omrane GUEDHAMI: Postprivatization Corporate
		Governance: the Role of Ownership Structure and Investor Protection
CLIM	38.2003	Rolf GOLOMBEK and Michael HOEL: Climate Policy under Technology Spillovers
KNOW	39.2003	Slim BEN YOUSSEF: Transboundary Pollution, R&D Spillovers and International Trade
CTN	40.2003	Carlo CARRARO and Carmen MARCHIORI: Endogenous Strategic Issue Linkage in International Negotiations
KNOW	41.2003	Sonia OREFFICE: Abortion and Female Power in the Household: Evidence from Labor Supply

(1) This paper was presented at the Workshop "Growth, Environmental Policies and Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, June 1, 2001

(li) This paper was presented at the Fourth Toulouse Conference on Environment and Resource Economics on "Property Rights, Institutions and Management of Environmental and Natural Resources", organised by Fondazione Eni Enrico Mattei, IDEI and INRA and sponsored by MATE, Toulouse, May 3-4, 2001

(lii) This paper was presented at the International Conference on "Economic Valuation of Environmental Goods", organised by Fondazione Eni Enrico Mattei in cooperation with CORILA, Venice, May 11, 2001

(liii) This paper was circulated at the International Conference on "Climate Policy – Do We Need a New Approach?", jointly organised by Fondazione Eni Enrico Mattei, Stanford University and Venice International University, Isola di San Servolo, Venice, September 6-8, 2001

(liv) This paper was presented at the Seventh Meeting of the Coalition Theory Network organised by the Fondazione Eni Enrico Mattei and the CORE, Université Catholique de Louvain, Venice, Italy, January 11-12, 2002

(lv) This paper was presented at the First Workshop of the Concerted Action on Tradable Emission Permits (CATEP) organised by the Fondazione Eni Enrico Mattei, Venice, Italy, December 3-4, 2001 (lvi) This paper was presented at the ESF EURESCO Conference on Environmental Policy in a Global Economy "The International Dimension of Environmental Policy", organised with the collaboration of the Fondazione Eni Enrico Mattei, Acquafredda di Maratea, October 6-11, 2001

(lvii) This paper was presented at the First Workshop of "CFEWE – Carbon Flows between Eastern and Western Europe", organised by the Fondazione Eni Enrico Mattei and Zentrum fur Europaische Integrationsforschung (ZEI), Milan, July 5-6, 2001

(lviii) This paper was presented at the Workshop on "Game Practice and the Environment", jointly organised by Università del Piemonte Orientale and Fondazione Eni Enrico Mattei, Alessandria, April 12-13, 2002

(Ivix) This paper was presented at the ENGIME Workshop on "Mapping Diversity", Leuven, May 16-17, 2002

(lvx) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by the Fondazione Eni Enrico Mattei, Milan, September 26-28, 2002

	2002 SERIES
CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
VOL	Voluntary and International Agreements (Editor: Carlo Carraro)
SUST	Sustainability Indicators and Environmental Valuation (Editor: Carlo Carraro)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Dino Pinelli)
MGMT	Corporate Sustainable Management (Editor: Andrea Marsanich)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)

2003 SERIES

CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
GG	Global Governance (Editor: Carlo Carraro)
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)
IEM	International Energy Markets (Editor: Anil Markandya)
CSRM	Corporate Social Responsibility and Management (Editor: Sabina Ratti)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)
CTN	Coalition Theory Network